

MEMORIAL

Journal Officiel du Grand-Duché de Luxembourg



MEMORIAL

Amtsblatt des Großherzogtums Luxemburg

RECUEIL DES SOCIETES ET ASSOCIATIONS

Le présent recueil contient les publications prévues par la loi modifiée du 10 août 1915 concernant les sociétés commerciales et par la loi modifiée du 21 avril 1928 sur les associations et les fondations sans but lucratif.

C — N° 1822

29 juillet 2013

SOMMAIRE

22Three Racing S.A 87426	Loke SA	. 87418
2512 H S.A 87411	Luxtom S.à r.l	87418
AB Distribution	Mainoria SICAV - SIF	87418
Albert & Partner S.A87410	MED Real Estates Venture Capital S.A	87411
A.N.D. INTERNATIONAL S.A., société de	Melbury JS S.à r.l	87418
gestion de patrimoine familial 87410	Miam Conseils S.A	87412
Anmaur Holdings (Luxembourg) Sàrl 87415	Moast S.A	87418
Asia Pacific Investment Holdings S.à r.l.	Mobilito s.à r.l	87418
87410	Montagu & Park Capital S.A	87419
Aveleos S.A	Montalbano S.A	87412
Aviation Finance and Leasing S.à r.l 87410	Novemife	87419
Banque BCP, S.A 87415	Novemiho	87419
Beauté CARA S.à r.l 87415	OICM Consulting	87419
Bopoz S.A87416	Oliwa S.A.	87412
Business Life Balancing Academy S.à r.l.	Omega Holding S.A	87419
	Opera S. à r. l	
Caros Investments S.A 87427	Oudenaarde Immobilier	
CDR Bounty (Luxembourg) S.à r.l 87427	Pentagon Holdings S.à r.l	87413
Central Europe Finance Immobiliare S.A.	Pfizer Europe Holdings Sarl	87413
87416	PK AirFinance	
CIH Holdings S.à r.l	Platinum UCITS Funds SICAV	87411
CIH International S.à r.l	Pomy S.à r.l	
Columbia Lux Capital SPF87416	PPInvest SICAV-SIF	
Costa Buena	Private Value A S.à r.l	87411
Costa Rei S.A	Private Value B S.à r.l	
Delille International S.A	Provencal Investments S.A	87414
Falconer II S.à r.l	Publifund	87414
Financière Poulain S.A	Riverland 2 S.à r.l	87420
Iris Participations et Investissements S.A.	Robeco Interest Plus Funds	87420
KDI Data il Dana anto 4 S à a l	Rosenthal Holding S.à r.l	
KPI Retail Property 1 S.à r.l	Rosenthal Holding S.à r.l.	
Larkas S.A	Simplify Partners S.A	
L.D.C. Partners S.à r.l	Soteman S.à r.l.	
Little Smets	Stam Rei III	



A.N.D. INTERNATIONAL S.A., société de gestion de patrimoine familial, Société Anonyme - Société de Gestion de Patrimoine Familial.

Siège social: L-2453 Luxembourg, 12, rue Eugène Ruppert.

R.C.S. Luxembourg B 4.064.

Le bilan et l'annexe au 31 décembre 2012 ainsi que les autres documents et informations qui s'y rapportent, ont été déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Pour A.N.D. INTERNATIONAL S.A., société de gestion de patrimoine familial

Signatures

Administrateur / Administrateur

Référence de publication: 2013075419/14.

(130093150) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Albert & Partner S.A., Société Anonyme.

Siège social: L-6630 Wasserbillig, 7, Grand-rue.

R.C.S. Luxembourg B 159.262.

Statuts coordonnés déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg.

Référence de publication: 2013075429/10.

(130093408) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Asia Pacific Investment Holdings S.à r.l., Société à responsabilité limitée.

Capital social: GBP 7.463.726,00.

Siège social: L-1249 Luxembourg, 2, rue du Fort Bourbon.

R.C.S. Luxembourg B 138.470.

Le bilan au 30 Septembre 2011 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 10 Juin 2013.

Référence de publication: 2013075445/10.

(130093542) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Aveleos S.A., Société Anonyme.

Siège social: L-1882 Luxembourg, 5, rue Guillaume Kroll.

R.C.S. Luxembourg B 153.427.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations. Luxembourg, le 30 mai 2013.

Référence de publication: 2013075450/10.

(130093228) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Aviation Finance and Leasing S.à r.l., Société à responsabilité limitée.

Siège social: L-2522 Luxembourg, 6, rue Guillaume Schneider.

R.C.S. Luxembourg B 168.032.

Les statuts coordonnés de la société ont été déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 18 février 2013.

Référence de publication: 2013075451/10.

(130093648) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



MED Real Estates Venture Capital S.A., Société Anonyme.

Siège social: L-2121 Luxembourg, 231, Val des Bons-Malades.

R.C.S. Luxembourg B 166.232.

Il résulte de la Résolution par écrit de l'actionnaire unique datée du 3 mai 2013 que la cooptation en date du 18 avril 2013 de Mme Maria Helena GONCALVES, née le 20 avril 1976 à Hayange (France), avec adresse professionnelle au 231, Val des Bons Malades, L-2121 Luxembourg, au poste d'Administrateur de la Société, en remplacement de Mme Geneviève BLAUEN-ARENDT, a été ratifiée.

Son mandat viendra à échéance à l'issue de l'assemblée générale qui se tiendra en 2017.

Pour extrait conforme SG AUDIT S.àr.l.

Référence de publication: 2013075161/14.

(130092434) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

2512 H S.A., Société Anonyme.

Siège social: L-1746 Luxembourg, 1, rue Joseph Hackin.

R.C.S. Luxembourg B 152.678.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Signature.

Référence de publication: 2013075372/10.

(130092925) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Platinum UCITS Funds SICAV, Société d'Investissement à Capital Variable.

Siège social: L-2346 Luxembourg, 20, rue de la Poste.

R.C.S. Luxembourg B 158.545.

Statuts coordonnés déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 10 juin 2013.

Référence de publication: 2013075210/10.

(130092352) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Private Value A S.à r.l., Société à responsabilité limitée.

Siège social: L-2449 Luxembourg, 16, boulevard Royal.

R.C.S. Luxembourg B 116.312.

Les comptes annuels au 31 Décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Signature.

Référence de publication: 2013075215/10.

(130092908) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Private Value B S.à r.l., Société à responsabilité limitée.

Siège social: L-2449 Luxembourg, 16, boulevard Royal.

R.C.S. Luxembourg B 116.311.

Les comptes annuels au 31 Décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Signature.

Référence de publication: 2013075216/10.

(130092907) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.



Oliwa S.A., Société Anonyme Soparfi.

Siège social: L-1660 Luxembourg, 60, Grand-rue.

R.C.S. Luxembourg B 64.000.

Extrait des résolutions prises lors de l'assemblée générale ordinaire tenue au siège social en date du 4 juin 2013

Fortunato CAUZZO démissionnaire du mandat d'Administrateur, est remplacée par MATRICE INTERNATIONAL SA, RCSL n° B 80362, 60 Grand-rue L-1660 Luxembourg.

Madeleine SARLETTE démissionnaire du mandat de commissaire aux comptes, est remplacée par Fortunato CAUZZO 60 Grand-rue L-1660 Luxembourg.

Ces mandats viendront à échéance lors de l'assemblée générale qui se tiendra en 2018.

Pour extrait sincère et conforme

EASIT S.A.

Référence de publication: 2013075198/15.

(130092231) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Miam Conseils S.A., Société Anonyme.

Siège social: L-1940 Luxembourg, 346A, route de Longwy.

R.C.S. Luxembourg B 105.522.

Il résulte des résolutions prises par l'actionnaire unique de la société en date du 7 juin 2013 que la société ODD Financial Services SA, ayant son siège social au 1-3, Boulevard de la Foire, L-1528 Luxembourg et enregistré au registre de commerce et de sociétés de Luxembourg sous le numéro RCS B41014 est nommée en remplacement Monsieur Bruno Cointepas, commissaire aux comptes dont le mandat arrive à échéance avec effet au 27 juin 2013 avec effet immédiat et ce pour une durée de 6 ans.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Fait à Luxembourg, le 10 juin 2013.

Référence de publication: 2013075164/14.

(130092511) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Montalbano S.A., Société Anonyme.

Siège social: L-1219 Luxembourg, 23, rue Beaumont.

R.C.S. Luxembourg B 45.381.

EXTRAIT

Madame Sandra Bortolus n'est plus administrateur depuis le 24 mai 2013.

Luxembourg, le 7 juin 2013.

Pour extrait conforme

Pour le conseil d'administration

Signature

Référence de publication: 2013075167/13.

(130092859) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

PK AirFinance, Société à responsabilité limitée.

Siège social: L-2633 Senningerberg, 6D, route de Trèves.

R.C.S. Luxembourg B 27.840.

Le bilan au 31 décembre 2012 a été déposé au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 10 Juin 2013.

Serge Michels

SVP Administration

Référence de publication: 2013075227/12.

(130092881) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.



Oudenaarde Immobilier, Société Anonyme.

Siège social: L-2522 Luxembourg, 6, rue Guillaume Schneider.

R.C.S. Luxembourg B 169.034.

Statuts coordonnés, suite un constat d'augmentation de capital reçu par Maître Francis KESSELER, notaire de résidence à Esch/Alzette, en date du 21 janvier 2013 déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Esch/Alzette, le 21 février 2013.

Francis KESSELER

NOTAIRE

Référence de publication: 2013075196/13.

(130092331) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Pentagon Holdings S.à r.l., Société à responsabilité limitée.

Siège social: L-1528 Luxembourg, 1-3, boulevard de la Foire.

R.C.S. Luxembourg B 153.017.

Statuts coordonnés, suite à l'assemblée générale extraordinaire reçue par Maître Francis KESSELER, notaire de résidence à Esch/Alzette, en date du 18 janvier 2013 déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Esch/Alzette, le 18 février 2013.

Francis KESSELER

NOTAIRE

Référence de publication: 2013075206/13.

(130092355) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Pfizer Europe Holdings Sàrl, Société à responsabilité limitée.

Siège social: L-1855 Luxembourg, 51, avenue J.F. Kennedy.

R.C.S. Luxembourg B 135.006.

Statuts coordonnés, suite à l'assemblée générale extraordinaire reçue par Maître Francis KESSELER, notaire de résidence à Esch/Alzette, en date du 31 décembre 2012 déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Esch/Alzette, le 31 janvier 2013.

Francis KESSELER

NOTAIRE

Référence de publication: 2013075225/13.

(130092485) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

PPInvest SICAV-SIF, Société Anonyme sous la forme d'une SICAV - Fonds d'Investissement Spécialisé.

Siège social: L-8217 Mamer, 41, Op Bierg.

R.C.S. Luxembourg B 164.706.

EXTRAIT

Le conseil d'administration a pris note de la démission de monsieur Philippe MELONI de son poste d'administrateur avec effet au 15 avril 2013.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Mamer, le 12 mai 2013.

Pour extrait conforme

Signature

Un mandataire

Référence de publication: 2013075231/15.

(130092830) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.



Provencal Investments S.A., Société Anonyme.

Siège social: L-2168 Luxembourg, 127, rue de Mühlenbach. R.C.S. Luxembourg B 116.230.

Les comptes annuels au 31 décembre 2011 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Signature.

Référence de publication: 2013075217/10.

(130092336) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Pomy S.à r.l., Société à responsabilité limitée.

Siège social: L-1653 Luxembourg, 2-8, avenue Charles de Gaulle.

R.C.S. Luxembourg B 79.947.

Le bilan au 31 décembre 2012 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 27 mai 2013.

Luxembourg Corporation Company S.A.

Signatures

Gérant

Référence de publication: 2013075230/13.

(130092661) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Publifund, Société d'Investissement à Capital Variable.

Siège social: L-1470 Luxembourg, 69, route d'Esch.

R.C.S. Luxembourg B 45.063.

Le Conseil d'Administration a pris note de la démission en tant qu'administrateur de:

Monsieur Jean-Michel LOEHR, 14, Porte de France, L-4360 Esch-sur-Alzette, en date du 1 er mai 2013.

Pour PUBLIFUND

Société d'Investissement à Capital Variable

RBC INVESTOR SERVICES BANK S.A.

Société Anonyme

Référence de publication: 2013075237/13.

(130092994) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Rosenthal Holding S.à r.l., Société à responsabilité limitée.

Capital social: USD 40.000,00.

Siège social: L-1653 Luxembourg, 2, avenue Charles de Gaulle.

R.C.S. Luxembourg B 88.950.

Les comptes annuels au 31 décembre 2011 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075259/9.

(130092745) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Rosenthal Holding S.à r.l., Société à responsabilité limitée.

Capital social: USD 40.000,00.

Siège social: L-1653 Luxembourg, 2, avenue Charles de Gaulle.

R.C.S. Luxembourg B 88.950.

Les comptes annuels au 31 décembre 2010 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075260/9.

(130092746) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.



Soteman S.à r.l., Société à responsabilité limitée.

Siège social: L-4437 Soleuvre, 196, rue de Differdange.

R.C.S. Luxembourg B 32.040.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075305/9.

(130092609) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Stam Rei III, Société en Commandite par Actions.

Siège social: L-1882 Luxembourg, 12F, rue Guillaume Kroll.

R.C.S. Luxembourg B 135.378.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075309/9.

(130092187) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Anmaur Holdings (Luxembourg) Sàrl, Société à responsabilité limitée unipersonnelle.

Siège social: L-5884 Hesperange, 300C, route de Thionville.

R.C.S. Luxembourg B 127.909.

Les comptes annuels au 31.12.2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075440/9.

(130093209) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

AB Distribution, Société à responsabilité limitée.

Siège social: L-8410 Steinfort, 58, route d'Arlon.

R.C.S. Luxembourg B 156.171.

Le Bilan au 31 décembre 2012 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075453/9.

(130093306) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Beauté CARA S.à r.l., Société à responsabilité limitée.

Siège social: L-2721 Luxembourg, 5, rue Alphonse Weicker.

R.C.S. Luxembourg B 104.317.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075491/9.

(130093274) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Banque BCP, S.A., Société Anonyme.

Siège social: L-8070 Bertrange, 5, rue des Mérovingiens, Z.A. Bourmicht.

R.C.S. Luxembourg B 7.648.

Statuts coordonnés déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075505/9.

(130093450) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



Bopoz S.A., Société Anonyme.

Siège social: L-1116 Luxembourg, 6, rue Adolphe.

R.C.S. Luxembourg B 143.752.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075521/9.

(130093813) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Business Life Balancing Academy S.à r.l., Société à responsabilité limitée.

Siège social: L-6475 Echternach, 20, Rabatt.

R.C.S. Luxembourg B 152.670.

Der Jahresabschluss vom 31.12.2012 wurde beim Handels- und Gesellschaftsregister von Luxemburg hinterlegt.

Zwecks Veröffentlichung im Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075526/9.

(130093171) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

CEFI, Central Europe Finance Immobiliare S.A., Société Anonyme.

Siège social: L-1724 Luxembourg, 3A, boulevard du Prince Henri.

R.C.S. Luxembourg B 114.274.

Les comptes annuels au 31/12/2011 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075561/9.

(130093917) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Columbia Lux Capital SPF, Société Anonyme - Société de Gestion de Patrimoine Familial.

Siège social: L-8308 Capellen, 75, Parc d'Activités.

R.C.S. Luxembourg B 148.049.

Les comptes annuels au 31/12/2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075580/9.

(130093142) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Costa Rei S.A., Société Anonyme.

Siège social: L-2165 Luxembourg, 26-28, Rives de Clausen.

R.C.S. Luxembourg B 133.027.

Les comptes annuels au 31 décembre 2010 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075590/9.

(130093591) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Delille International S.A., Société Anonyme.

Siège social: L-1724 Luxembourg, 3A, boulevard du Prince Henri.

R.C.S. Luxembourg B 167.406.

Les comptes annuels au 31/12/2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075617/9.

(130093923) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



Financière Poulain S.A., Société Anonyme.

Siège social: L-1724 Luxembourg, 3A, boulevard du Prince Henri.

R.C.S. Luxembourg B 167.731.

Les comptes annuels au 31/12/2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075665/9.

(130093924) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Falconer II S.à r.l., Société à responsabilité limitée.

Siège social: L-2633 Senningerberg, 6A, route de Trèves.

R.C.S. Luxembourg B 147.530.

Le bilan au 31 décembre 2011 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075668/9.

(130093342) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

KPI Retail Property 1 S.à r.l., Société à responsabilité limitée.

Capital social: EUR 12.500,00.

Siège social: L-2180 Luxembourg, 6, rue Jean Monnet.

R.C.S. Luxembourg B 108.112.

Les comptes annuels au 30 juin 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075790/9.

(130093673) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Larkas S.A., Société Anonyme.

Siège social: L-5365 Munsbach, 22, rue Gabriel Lippmann.

R.C.S. Luxembourg B 44.369.

Les comptes annuels au 31 décembre 2011 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075818/9.

(130093199) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Little Smets, Société à responsabilité limitée.

Siège social: L-8010 Strassen, 262, route d'Arlon.

R.C.S. Luxembourg B 24.325.

Les comptes annuels au 31/12/2011 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075831/9.

(130093807) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Logiconsult SA, Société Anonyme.

Siège social: L-3285 Bettembourg, 22, rue Willmar.

R.C.S. Luxembourg B 82.165.

Les comptes annuels au 31.12.2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations. Référence de publication: 2013075833/9.

(130093208) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



Loke SA, Société Anonyme.

Siège social: L-1720 Luxembourg, 6, rue Heinrich Heine.

R.C.S. Luxembourg B 76.303.

Les comptes annuels au 31.12.2012 ont été déposés au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075834/9.

(130093758) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Luxtom S.à r.l., Société à responsabilité limitée.

Siège social: L-8009 Strassen, 155A, route d'Arlon.

R.C.S. Luxembourg B 146.623.

Le Bilan au 31 janvier 2013 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075848/9.

(130093308) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Mainoria SICAV - SIF, Société d'Investissement à Capital Variable - Fonds d'Investissement Spécialisé.

Siège social: L-1142 Luxembourg, 1A, rue Pierre d'Aspelt.

R.C.S. Luxembourg B 35.627.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075858/9.

(130093860) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Melbury JS S.à r.l., Société à responsabilité limitée.

Siège social: L-2633 Senningerberg, 6A, route de Trèves.

R.C.S. Luxembourg B 129.013.

Le bilan au 31 décembre 2011 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075872/9.

(130093373) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Moast S.A., Société Anonyme.

Siège social: L-1835 Luxembourg, 13, rue des Jardiniers.

R.C.S. Luxembourg B 145.966.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075882/9.

(130093255) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Mobilito s.à r.l., Société à responsabilité limitée.

Siège social: L-8437 Steinfort, 60, rue de Koerich.

R.C.S. Luxembourg B 139.582.

Le Bilan au 31 décembre 2012 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075884/9.

(130093305) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



Montagu & Park Capital S.A., Société Anonyme.

Siège social: L-2128 Luxembourg, 22, rue Marie-Adélaïde.

R.C.S. Luxembourg B 131.673.

Le bilan au 31 décembre 2010 a été déposé au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075885/9.

(130093343) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Novemife, Société à responsabilité limitée.

Capital social: EUR 12.500,00.

Siège social: L-1116 Luxembourg, 6, rue Adolphe.

R.C.S. Luxembourg B 163.072.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075897/9.

(130093752) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Novemiho, Société à responsabilité limitée.

Capital social: EUR 12.500,00.

Siège social: L-1116 Luxembourg, 6, rue Adolphe.

R.C.S. Luxembourg B 145.342.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075898/9.

(130093751) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

OICM Consulting, Société Anonyme.

Siège social: L-2723 Howald, 72, rue Eugène Welter.

R.C.S. Luxembourg B 146.113.

Les comptes annuels au 31 décembre 2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075908/9.

(130093495) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Omega Holding S.A., Société Anonyme.

Siège social: L-2453 Luxembourg, 2-4, rue Eugène Ruppert.

R.C.S. Luxembourg B 167.588.

Les comptes annuels au 31.12.2012 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075909/9.

(130093684) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

Opera S. à r. l., Société à responsabilité limitée.

Siège social: L-1653 Luxembourg, 2, avenue Charles de Gaulle.

R.C.S. Luxembourg B 131.953.

Les comptes annuels au 31 décembre 2010 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Référence de publication: 2013075911/9.

(130093811) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



Riverland 2 S.à r.l., Société à responsabilité limitée.

Siège social: L-1855 Luxembourg, 46A, avenue J.F. Kennedy.

R.C.S. Luxembourg B 148.405.

Les comptes annuels au 31 décembre 2011 ont été déposés au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 4 juin 2013.

Riverland 2 S.à r.l.

G.B.AD. Cousin / P.L.C. van Denzen

Gérant / Gérant

Référence de publication: 2013075245/13.

(130092828) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Robeco Interest Plus Funds, Société d'Investissement à Capital Variable.

Siège social: L-1470 Luxembourg, 69, route d'Esch.

R.C.S. Luxembourg B 40.490.

Le rapport annuel au 31 décembre 2012 a été déposé au registre de commerce et des sociétés de Luxembourg. Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

ROBECO INTEREST PLUS FUNDS

Société d'Investissement à Capital Variable

RBC INVESTOR SERVICES BANK S.A.

Société Anonyme

Référence de publication: 2013075256/13.

(130092844) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Simplify Partners S.A., Société Anonyme.

Siège social: L-2449 Luxembourg, 25A, boulevard Royal.

R.C.S. Luxembourg B 140.483.

Société anonyme fondée le 09 juin 2008 et publication dans le Mémorial C-N° 2000.

Les comptes annuels de 2011 ont été clôturés au 31 Décembre 2011 et approuvés lors de l'assemblée ordinaire des actionnaires le 15 mai 2012 au registre de commerce et des sociétés de Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 14/06/2012.

Paddock Fund Administration

Référence de publication: 2013075269/13.

(130092425) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Iris Participations et Investissements S.A., Société Anonyme.

R.C.S. Luxembourg B 83.168.

CLÔTURE DES LIQUIDATIONS

Par jugements rendus en date du 6 juin 2013, le tribunal d'arrondissement de Luxembourg, 6 ^{ème} chambre, siégeant en matière commerciale, après avoir entendu le juge commissaire en son rapport oral, le liquidateur et le Ministère Public en leurs conclusions, a déclaré closes pour absence d'actif les opérations de liquidation des sociétés suivantes:

Ces mêmes jugements ont mis les frais à la charge du Trésor.

Luxembourg, le 10 juin 2013.

Pour extrait conforme

Me Sevinc GUVENCE

Le liquidateur

Référence de publication: 2013075405/15.

(130092764) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.



CIH Holdings S.à r.l., Société à responsabilité limitée.

Capital social: USD 25.050,00.

Siège social: L-1882 Luxembourg, 5, rue Guillaume Kroll. R.C.S. Luxembourg B 176.841.

In the year two thousand thirteen, on the twenty-eight day of May,

Before Us, Maître Martine SCHAEFFER, notary residing in Luxembourg, Grand Duchy of Luxembourg, in place of Maître Joseph ELVINGER, notary residing in Luxembourg, currently prevented, who will guard the original of the present deed.

THERE APPEARED

Constellation International CWI Holdings S.C.S., a limited partnership (société en commandite simple) incorporated under the laws of the Grand Duchy of Luxembourg, having its registered office at 9A, rue Robert Stümper, L-2557 Luxembourg and registered with the Luxembourg Register of Commerce and Companies under number B 176.604, acting through its general partner CWI Holdings LLC, a New York limited liability company incorporated and organized under the laws of the State of New York, with registered office at 207, High Point Drive, Bldg. 100, Victor, New York 14564 and registered with the Office of the Secretary of State of the New York Department of State under number 3472881,

here represented by Mrs Rachel UHL, lawyer, professionally residing in Luxembourg by virtue of a proxy given under private seal, which, signed ne varietur by the appearing party and the undersigned notary, will remain attached to the present deed to be filed at the same time with the registration authorities.

Such appearing party is the sole shareholder (the "Sole Shareholder") of CIH HOLDINGS S.à r.l., a private limited liability company (société à responsabilité limitée) incorporated under the laws of the Grand Duchy of Luxembourg, having its registered office at 5, rue Guillaume Kroll, L-1882 Luxembourg, registered with the Luxembourg Register of Commerce and Companies under number B 176.841 (the "Company"), incorporated pursuant to a deed of Maître Joseph ELVINGER, notary public, residing in Luxembourg, Grand Duchy of Luxembourg, dated 15 April 2013, not yet published in the Mémorial C, Recueil des Sociétés et Associations, whose articles of association (the "Articles") have not been amended since its incorporation.

The Sole Shareholder, representing the whole corporate capital of the Company and represented as indicated above, requires the notary to act the following resolutions taken in accordance with the provisions of article 200-2 of the Luxembourg law on commercial companies of 10 August 1915, as amended, pursuant to which a sole shareholder of a société à responsabilité limitée shall exercise the powers of the general meeting of shareholders of the Company and the decisions of the sole shareholder are recorded in minutes or drawn up in writing:

First resolution

The Sole Shareholder resolves to increase the share capital of the Company by an amount of fifty US Dollars (USD 50.-), in order to raise it from its current amount of twenty-five thousand US Dollars (USD 25,000.-) to twenty-five thousand and fifty US Dollars (USD 25,050.-), through the creation and issuance of one (1) new share, with a nominal value of fifty US Dollars (USD 50.-) (the "New Share"), having the same rights and obligations as the existing shares, to be issued and fully paid up with a share premium for an amount of one million nineteen thousand nine hundred and fifty US Dollars (USD 1,019,950.-) (the "Share Premium").

Subscription and Payment

The Sole Shareholder declares to subscribe for the ownership of the New Share and to fully pay up such New Share, together with the Share Premium, by means of a contribution in cash for a total amount of one million twenty thousand US Dollars (USD 1,020,000.-).

The amount of one million twenty thousand US Dollars (USD 1,020,000.-) is now at the disposal of the Company, proof of which has been duly given to the undersigned notary.

Fifty US Dollars (USD 50.-) shall be allocated to the Company's share capital account and one million nineteen thousand nine hundred and fifty US Dollars (USD 1,019,950.-) to the Company's share premium account.

Second resolution

As a consequence of the above resolution, the Sole Shareholder resolves to amend the first paragraph of article 5 of the Articles, which shall henceforth read as follows:

"The share capital is fixed at USD 25,050.-(twenty-five thousand and fifty US Dollars) represented by 501 (five hundred and one) shares of USD 50.-(fifty US Dollars) each."

Costs and Expenses

The expenses, costs, remunerations or charges in any form whatsoever which shall be borne by the Company as a result of the present deed are estimated at approximately two thousand euros.



Whereof, the present deed is drawn up in Luxembourg, on the day stated at the beginning of this document.

The undersigned notary who understands and speaks English, states herewith that at the request of the appearing party, the present deed is worded in English, followed by a French version and that in case of divergences between the English and the French texts, the English version will be preponderant.

The document having been read to the proxyholder of the appearing party, known to the notary by name, first name, civil status and residence, said proxyholder of the appearing party signed together with the notary the present deed.

Suit la traduction française de ce qui précède:

L'an deux mille treize, le vingt-huit mai,

Par-devant Maître Martine SCHAEFFER, notaire de résidence à Luxembourg, en remplacement de Maître Joseph ELVINGER, notaire de résidence à Luxembourg, Grand-Duché de Luxembourg, actuellement empêché, lequel aura la garde de la présente minute.

A COMPARU

Constellation International CWI Holdings S.C.S., une société en commandite simple constituée selon les lois du Grand-Duché de Luxembourg, ayant son siège social au 9A, rue Robert Stümper, L-2557 Luxembourg et immatriculée auprès du Registre de Commerce et des Sociétés de Luxembourg sous le numéro B 176.604, agissant par son general partner CWI Holdings LLC, une limited liability company constituée selon les lois de l'Etat de New York, ayant son siège social au 207, High Point Drive, Bldg. 100, Victor, New York 14564 et immatriculée auprès du Office of the Secretary of State of the New York Department sous le numéro 3472881,

ici représenté par Madame Rachel UHL, juriste, résidant professionnellement à Luxembourg, en vertu d'une procuration sous seing privé, laquelle, signée ne varietur par la partie comparante et par le notaire instrumentant restera annexée au présent acte pour être soumise avec lui aux formalités de l'enregistrement.

Laquelle partie comparante est l'associé unique (l'"Associé Unique") de CIH HOLDINGS S.à r.l., une société à responsabilité limitée constituée selon les lois du Grand-Duché de Luxembourg, ayant son siège social au 5, rue Guillaume Kroll, L-1882 Luxembourg, immatriculée auprès du Registre de Commerce et des Sociétés de Luxembourg sous le numéro B 176.841 (la "Société"), constituée par un acte rédigé par Maître Joseph ELVINGER, notaire de résidence à Luxembourg, Grand-Duché de Luxembourg, en date du 15 avril 2013, non encore publié au Mémorial C, Recueil des Sociétés et Association, dont les statuts (les "Statuts") n'ont pas encore été modifiés depuis la constitution.

l'Associé Unique, représentant l'intégralité du capital social de la Société et représenté comme indiqué ci-dessus, requiert le notaire d'acter les résolutions suivantes prises conformément aux dispositions de l'article 200-2 de la loi luxembourgeoise sur les sociétés commerciales du 10 août 1915, telle que modifiée, en vertu duquel un associé unique d'une société à responsabilité limitée exercera les pouvoirs dévolus à l'assemblée générale des associés de la Société et les décisions de l'associé unique seront actées dans des procès-verbaux ou prises par écrit:

Première résolution

L'Associé Unique décide d'augmenter le capital de la Société d'un montant de cinquante Dollars Américains (50,-USD), afin de l'augmenter de son montant actuel de vingt-cinq mille Dollars Américains (25.000,-USD) à vingt-cinq mille cinquante Dollars Américains (25.050,USD), par la création et l'émission d'une (1) nouvelle part sociale, avec une valeur nominale de cinquante Dollars Américains (50,-USD) (la "Nouvelle Part Sociale"), ayant les mêmes droits et obligations que les parts sociales existantes, devant être émise et entièrement libérée avec une prime d'émission d'un montant d'un million dix-neuf mille neuf cent cinquante Dollars Américains (1.019.950,-USD) (la "Prime d'Emission")].

Souscription et Paiement

L'Associé Unique déclare souscrire à la Nouvelle Part Sociale pour un montant total d'un million vingt mille Dollars Américains (1.020.000,-USD), et entièrement libérer cette Nouvelle Part Sociale, avec la Prime d'Emission, par un apport en numéraire.

La somme d'un million vingt mille Dollars Américains (1.020.000,-USD) se trouve désormais à la disposition de la Société, ainsi qu'il en a été justifié au notaire instrumentant.

Cinquante Dollars Américains (50,-USD) sont alloués au capital social de la Société et un million dix-neuf mille neuf cent cinquante Dollars Américains (1.019.950,-USD) sont alloués au compte de prime d'émission de la Société].

Deuxième résolution

A la suite de la résolution ci-dessus, l'Associé Unique décide de modifier le premier paragraphe de l'article 5 des Statuts, qui sera désormais la teneur suivante:

"Le capital social est fixé à 25.050,-Dollars Américains (vingt-cinq mille cinquante Dollars Américains) représenté par 501 (cinq cent un) parts sociales d'une valeur nominale de 50,-Dollars Américains (cinquante Dollars Américains) chacune."



Frais et Dépenses

Les frais, dépenses, rémunérations et charges de quelque forme que ce soit qui incombent à la société en raison des présentes, est évalué à environ deux mille euros.

Dont acte, fait et passé à Luxembourg, le jour mentionné en haut de ce document.

Le notaire instrumentant qui comprend et parle l'anglais, déclare qu'à la demande de la partie comparante, le présent acte est rédigé en anglais suivi par une version française et qu' en cas de divergence entre le texte français et le texte anglais, le texte anglais fera foi.

Le document a été lu au mandataire de la partie comparante, connu du notaire par son nom, prénom, état civil et domicile, lequel mandataire du comparant a signé avec le notaire le présent acte.

Signé: R. UHL, M. SCHAEFFER.

Enregistré à Luxembourg Actes Civils le 7 juin 2013. Relation: LAC/2013/25963.

Reçu soixante quinze euros (EUR 75,-).

Le Receveur (signé): C. FRISING.

Référence de publication: 2013075571/123.

(130093713) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

CIH International S.à r.l., Société à responsabilité limitée.

Capital social: USD 25.050,00.

Siège social: L-1882 Luxembourg, 5, rue Guillaume Kroll.

R.C.S. Luxembourg B 176.850.

In the year two thousand thirteen, on the twenty-eight day of May,

Before Us, Maître Martine SCHAEFFER, notary residing in Luxembourg, Grand Duchy of Luxembourg, in place of Maître Joseph ELVINGER, notary residing in Luxembourg, currently prevented, who will guard the original of the present deed.

THERE APPEARED

CIH HOLDINGS S.à r.l., a private limited liability company (société à responsabilité limitée) incorporated under the laws of the Grand Duchy of Luxembourg, having its registered office at 5, rue Guillaume Kroll, L-1882 Luxembourg and registered with the Luxembourg Register of Commerce and Companies under number B 176.841,

here represented by Mrs Rachel UHL, lawyer, professionally residing in Luxembourg by virtue of a proxy given under private seal, which, signed ne varietur by the appearing party and the undersigned notary, will remain attached to the present deed to be filed at the same time with the registration authorities.

Such appearing party is the sole shareholder (the "Sole Shareholder") of CIH INTERNATIONAL S.à r.l., a private limited liability company (société à responsabilité limitée) incorporated under the laws of the Grand Duchy of Luxembourg, having its registered office at 5, rue Guillaume Kroll, L-1882 Luxembourg, registered with the Luxembourg Register of Commerce and Companies under number B 176.850 (the "Company"), incorporated pursuant to a deed of Maître Joseph ELVINGER, notary public, residing in Luxembourg, Grand Duchy of Luxembourg, dated 15 April 2013, not yet published in the Mémorial C, Recueil des Sociétés et Associations, whose articles of association (the "Articles") have not been amended since its incorporation.

The Sole Shareholder, representing the whole corporate capital of the Company and represented as indicated above, requires the notary to act the following resolutions taken in accordance with the provisions of article 200-2 of the Luxembourg law on commercial companies of 10 August 1915, as amended, pursuant to which a sole shareholder of a société à responsabilité limitée shall exercise the powers of the general meeting of shareholders of the Company and the decisions of the sole shareholder are recorded in minutes or drawn up in writing:

First resolution

The Sole Shareholder resolves to increase the share capital of the Company by an amount of fifty US Dollars (USD 50.-), in order to raise it from its current amount of twenty-five thousand US Dollars (USD 25,000.-) to twenty-five thousand and fifty US Dollars (USD 25,050.-), through the creation and issuance of one (1) new share, with a nominal value of fifty US Dollars (USD 50.-) (the "New Share"), having the same rights and obligations as the existing shares, to be issued and fully paid up with a share premium for an amount of one million nineteen thousand nine hundred and fifty US Dollars (USD 1,019,950.-) (the "Share Premium").

Subscription and Payment

The Sole Shareholder declares to subscribe for the ownership of the New Share and to fully pay up such New Share, together with the Share Premium, by means of a contribution in cash for a total amount of one million twenty thousand US Dollars (USD 1,020,000.-).



The amount of one million twenty thousand US Dollars (USD 1,020,000.-) is now at the disposal of the Company, proof of which has been duly given to the undersigned notary.

Fifty US Dollars (USD 50.-) shall be allocated to the Company's share capital account and one million nineteen thousand nine hundred and fifty US Dollars (USD 1,019,950.-) to the Company's share premium account.

Second resolution

As a consequence of the above resolution, the Sole Shareholder resolves to amend the first paragraph of article 5 of the Articles, which shall henceforth read as follows:

"The share capital is fixed at USD 25,050.-(twenty-five thousand and fifty US Dollars) represented by 501 (five hundred and one) shares of USD 50.-(fifty US Dollars) each."

Costs and Expenses

The expenses, costs, remunerations or charges in any form whatsoever which shall be borne by the Company as a result of the present deed are estimated at approximately two thousand euros.

Whereof, the present deed is drawn up in Luxembourg, on the day stated at the beginning of this document.

The undersigned notary who understands and speaks English, states herewith that at the request of the appearing party, the present deed is worded in English, followed by a French version and that in case of divergences between the English and the French texts, the English version will be preponderant.

The document having been read to the proxyholder of the appearing party, known to the notary by name, first name, civil status and residence, said proxyholder of the appearing party signed together with the notary the present deed.

Suit la traduction française de ce qui précède:

L'an deux mille treize, le vingt-huit mai,

Par-devant Maître Martine SCHAEFFER, notaire de résidence à Luxembourg, en remplacement de Maître Joseph ELVINGER, notaire de résidence à Luxembourg, Grand-Duché de Luxembourg, actuellement empêché, lequel aura la garde de la présente minute.

A COMPARU

CIH HOLDINGS S.à r.l., une société à responsabilité limitée constituée selon les lois du Grand-Duché de Luxembourg, ayant son siège social au 5, rue Guillaume Kroll, L-1882 Luxembourg et immatriculée auprès du Registre de Commerce et des Sociétés de Luxembourg sous le numéro B 176.841,

ici représenté par Madame Rachel UHL, juriste, résidant professionnellement à Luxembourg, en vertu d'une procuration sous seing privé, laquelle, signée ne varietur par la partie comparante et par le notaire instrumentant restera annexée au présent acte pour être soumise avec lui aux formalités de l'enregistrement.

Laquelle partie comparante est l'associé unique (l'"Associé Unique") de CIH INTERNATIONAL S.à r.l., une société à responsabilité limitée constituée selon les lois du Grand-Duché de Luxembourg, ayant son siège social au 5, rue Guillaume Kroll, L-1882 Luxembourg, immatriculée auprès du Registre de Commerce et des Sociétés de Luxembourg sous le numéro B 176.850 (la "Société"), constituée par un acte rédigé par Maître Joseph ELVINGER, notaire de résidence à Luxembourg, Grand-Duché de Luxembourg, en date du 15 avril 2013, non encore publié au Mémorial C, Recueil des Sociétés et Association, dont les statuts (les "Statuts") n'ont pas encore été modifiés depuis la constitution.

l'Associé Unique, représentant l'intégralité du capital social de la Société et représenté comme indiqué ci-dessus, requiert le notaire d'acter les résolutions suivantes prises conformément aux dispositions de l'article 200-2 de la loi luxembourgeoise sur les sociétés commerciales du 10 août 1915, telle que modifiée, en vertu duquel un associé unique d'une société à responsabilité limitée exercera les pouvoirs dévolus à l'assemblée générale des associés de la Société et les décisions de l'associé unique seront actées dans des procès-verbaux ou prises par écrit:

Première résolution

L'Associé Unique décide d'augmenter le capital de la Société d'un montant de cinquante Dollars Américains (50,-USD), afin de l'augmenter de son montant actuel de vingt-cinq mille Dollars Américains (25.000,-USD) à vingt-cinq mille cinquante Dollars Américains (25.050,USD), par la création et l'émission d'une (1) nouvelle part sociale, avec une valeur nominale de cinquante Dollars Américains (50,-USD) (la "Nouvelle Part Sociale"), ayant les mêmes droits et obligations que les parts sociales existantes, devant être émise et entièrement libérée avec une prime d'émission d'un montant d'un million dix-neuf mille neuf cent cinquante Dollars Américains (1.019.950,-USD) (la "Prime d'Emission")].

Souscription et Paiement

L'Associé Unique déclare souscrire à la Nouvelle Part Sociale pour un montant total d'un million vingt mille Dollars Américains (1.020.000,-USD), et entièrement libérer cette Nouvelle Part Sociale, avec la Prime d'Emission, par un apport en numéraire.

La somme d'un million vingt mille Dollars Américains (1.020.000,-USD) se trouve désormais à la disposition de la Société, ainsi qu'il en a été justifié au notaire instrumentant.



Cinquante Dollars Américains (50,-USD) sont alloués au capital social de la Société et un million dix-neuf mille neuf cent cinquante Dollars Américains (1.019.950,-USD) sont alloués au compte de prime d'émission de la Société].

Deuxième résolution

A la suite de la résolution ci-dessus, l'Associé Unique décide de modifier le premier paragraphe de l'article 5 des Statuts, qui sera désormais la teneur suivante:

"Le capital social est fixé à 25.050,-Dollars Américains (vingt-cinq mille cinquante Dollars Américains) représenté par 501 (cinq cent un) parts sociales d'une valeur nominale de 50,-Dollars Américains (cinquante Dollars Américains) chacune."

Frais et Dépenses

Les frais, dépenses, rémunérations et charges de quelque forme que ce soit qui incombent à la société en raison des présentes, est évalué à environ deux mille euros.

Dont acte, fait et passé à Luxembourg, le jour mentionné en haut de ce document.

Le notaire instrumentant qui comprend et parle l'anglais, déclare qu'à la demande de la partie comparante, le présent acte est rédigé en anglais suivi par une version française et qu' en cas de divergence entre le texte français et le texte anglais, le texte anglais fera foi.

Le document a été lu au mandataire de la partie comparante, connu du notaire par son nom, prénom, état civil et domicile, lequel mandataire du comparant a signé avec le notaire le présent acte.

Signé: R. UHL, M. SCHAEFFER.

Enregistré à Luxembourg Actes Civils le 7 juin 2013. Relation. LAC/2013/25964.

Reçu soixante quinze euros (EUR 75,-).

Le Receveur (signé): C. FRISING.

Référence de publication: 2013075573/117.

(130093750) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

L.D.C. Partners S.à r.l., Société à responsabilité limitée.

Siège social: L-9153 Dirbach, 5, Rannerbach. R.C.S. Luxembourg B 170.854.

DISSOLUTION

L'an deux mille treize, le cinq juin,

par-devant Maître Jean SECKLER, notaire de résidence à Junglinster, Grand-Duché de Luxembourg.

Ont comparu:

- 1.- Madame Christelle LONGAGNANI, sans profession, née à Villerupt (France), le 8 janvier 1972, demeurant à F-54400 Longwy, 36, avenue du 8 mai 1945;
- 2.- Monsieur David LONGAGNANI, commercial, né à Villerupt (France), le 8 août 1970, demeurant à F-57130 Arssur-Moselle, 5, Allée des Roses;
- 3.- Monsieur Léon SCHLECHTER, retraité, né à Heisdorf (Luxembourg), le 3 février 1943, demeurant à L-7371 Helmdange, 10, rue J.-F. Kennedy,

(ensemble ci-après les «comparants», ou encore les «associés»).

Lesquels comparants ont déclaré être les seuls propriétaires de l'intégralité du capital social de la société à responsabilité limitée L.D.C. Partners S.à r.l., avec siège social au 5, Rannerbach, L-9153 Dirbach, et ont requis le notaire instrumentant de documenter ce qui suit:

- 1. Que la société L.D.C. Partners S.à r.l., société à responsabilité limitée, R.C.S. Luxembourg B170854, ayant son siège social au 15, Rannerbach, L-9153 Dirbach, a été constituée le 9 juillet 2012 suivant acte reçu par le notaire instrumentant, publié au Mémorial C, Recueil des Sociétés et Associations n° 2297 du 14 septembre 2012.
- 2. Que le capital social de la société L.D.C. Partners S.à r.l. s'élève actuellement à EUR 12.600,- (douze mille six cents euros) représenté par 126 (cent vingt-six) parts sociales d'une valeur nominale de EUR 100,- (cent euros) chacune, entièrement libérées, et souscrites comme suit:

Madame Christelle LONGAGNANI:42 parts sociales,Monsieur David LONGAGNANI:42 parts sociales,Monsieur Léon SCHLECHTER:42 parts sociales,

- 3. Que par la présente, les associés, en tant que propriétaires de l'intégralité des parts sociales, prononcent la dissolution anticipée de la société avec effet immédiat.
- 4. Que les associés connaissent parfaitement la situation financière de la société et déclarent que l'activité de la société a cessé.



- 5. Que les associés déclarent que toutes les dettes connues ont été payées, que la liquidation de la société a été effectuée volontairement par accord à l'amiable, aux droits de toutes les parties qui pourraient se trouver concernées, et qu'ils prennent à leur charge tous les actifs, passifs et engagements financiers de la société dissoute, solidairement et indivisiblement vis-à-vis des tiers et chacun d'eux proportionnellement au nombre de parts qu'il possède dans leurs rapports entre eux.
 - 6. Que la liquidation est achevée et clôturée.
 - 7. Que les associés donnent décharge pleine et entière aux gérants de la société.
 - 8. Que le mandataire ou le notaire instrumentant peuvent procéder à l'annulation du registre de la société.
- 9. Que les livres et documents de la société dissoute seront conservés pendant cinq ans à l'adresse du siège social de la société.

Frais

Le montant des frais, dépenses, rémunérations et charges, sous quelque forme que ce soit, qui incombent à la Société à raison de cet acte, est dès lors évalué à neuf cents euros (EUR 900,-).

DONT ACTE, fait et passé à Junglinster, date qu'en tête des présentes.

Et après lecture faite aux comparants, connus du notaire instrumentant par noms, prénoms, états et demeures, ils ont signé avec Nous, notaire, la présente minute.

Signé: Christelle LONGAGNANI, David LONGAGNANI, Léon SCHLECHTER, Jean SECKLER.

Enregistré à Grevenmacher, le 07 juin 2013. Relation GRE/2013/2294. Reçu soixante-quinze euros 75,00 €.

Le Receveur (signé): G. SCHLINK.

Référence de publication: 2013075793/53.

(130092922) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

22Three Racing S.A., Société Anonyme.

Siège social: L-2661 Luxembourg, 42, rue de la Vallée.

R.C.S. Luxembourg B 160.607.

L'assemblée générale ordinaire du 10 juin 2013 a ratifié la décision prise par le Conseil d'Administration par voie circulaire de coopter Monsieur Yves BIEWER au poste d'administrateur de catégorie A de la société, en remplacement de Madame Marie BOURLOND démissionnaire. Le mandat de l'administrateur définitivement élu s'achèvera avec ceux des autres Administrateurs à l'issue de l'assemblée générale annuelle de 2016.

Lors de cette même assemblée générale ordinaire du 10 juin 2013, Monsieur Giuliano BIDOLI, 42, rue de la Vallée, L-2661 Luxembourg a été nommé administrateur de catégorie A au Conseil d'Administration avec effet rétroactif au 05 décembre 2012 en remplacement de Monsieur Olivier LECLIPTEUR démissionnaire.

Son mandat s'achèvera avec ceux des autres Administrateurs à l'issue de l'assemblée générale annuelle de 2016.

Luxembourg, le 10/06/2013.

Pour: 22THREE RACING S.A.

Société anonyme

Experta Luxembourg

Société anonyme

Cindy Szabo / Isabelle Marechal-Gerlaxhe

Référence de publication: 2013075370/21.

(130092684) Déposé au registre de commerce et des sociétés de Luxembourg, le 10 juin 2013.

Costa Buena, Société Anonyme.

Siège social: L-2180 Luxembourg, 6, rue Jean Monnet.

R.C.S. Luxembourg B 93.766.

Par décisions écrites du 7 juin 2013 le conseil d'administration a décidé de transférer le siège social de la société de 11a bld Joseph II, L-1840 Luxembourg à 6, Rue Jean Monnet, L-2180 Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Pour Costa Buena S.A.

Un mandataire

Référence de publication: 2013075588/12.

(130093069) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.



Caros Investments S.A., Société Anonyme.

Siège social: L-2449 Luxembourg, 25C, boulevard Royal.

R.C.S. Luxembourg B 85.358.

Il résulte du procès-verbal de l'Assemblée Générale Extraordinaire de la société tenue en date du 1 er juin 2013, que:

- Le siège social de la société a été transféré, avec effet immédiat, du 18, rue Robert Stümper, L-2557 Luxembourg au 25C, Boulevard Royal, L-2449 Luxembourg.

Pour mention aux fins de la publication au Mémorial, Recueil des Sociétés et Associations.

Luxembourg, le 7 juin 2013.

Pour extrait conforme

Signature

Un mandataire

Référence de publication: 2013075553/15.

(130093661) Déposé au registre de commerce et des sociétés de Luxembourg, le 11 juin 2013.

CDR Bounty (Luxembourg) S.à r.l., Société à responsabilité limitée.

Capital social: GBP 1.989.166,73.

Siège social: L-1882 Luxembourg, 5, rue Guillaume Kroll.

R.C.S. Luxembourg B 173.461.

In the year two thousand thirteen, on the twenty-fourth day of May,

Before Me Francis KESSELER, notary public, residing in Esch/Alzette (Grand Duchy of Luxembourg), undersigned,

Was held an extraordinary general meeting of the shareholders of CDR Bounty (Luxembourg) S.à r.l., a société à responsabilité limitée incorporated under the laws of the Grand Duchy of Luxembourg, having its registered office at 5, rue Guillaume Kroll, L-1882 Luxembourg, Grand Duchy of Luxembourg, registered with the Luxembourg Trade and Companies Register under number B 173.461 (the "Company"), incorporated pursuant to a deed of Maître Francis KES-SELER, notary public, residing in Esch/Alzette (Grand Duchy of Luxembourg), dated 27 November 2012, published in the Mémorial C, Recueil des Sociétés et Associations (the "Mémorial") number 203, page 9717 dated 28 January 2013. The articles of association of the Company (the "Articles") have been amended for the last time by a notarial deed of Maître Francis KESSELER, notary, residing in Esch/Alzette (Grand Duchy of Luxembourg), dated 6 March 2013, not yet published in the Mémorial.

The Meeting elected Mrs Sofia AFONSO-DA CHAO CONDE, private employee, residing professionally in Esch/Alzette, as chairman of the Meeting (the "Chairman"). The Chairman appointed Mrs Claudia ROUCKERT, private employee, residing professionally in Esch/Alzette, as secretary of the Meeting (the "Secretary"). The Meeting elected Mrs Maria SANTIAGO-DE SOUSA, private employee, residing professionally in Esch/Alzette, as scrutineer of the Meeting (the "Scrutineer").

These appointments having been made, the Chairman declared that and requested the notary to state that:

- I. The shareholders represented and the number of shares held by them are indicated on an attendance list (the "Shareholders"). This list and the proxies, after having been signed by the proxyholder(s) of the Shareholders and the undersigned notary, will remain attached to the present deed for registration purposes.
- II. This attendance list shows that 198,916,673 shares of the Company, representing 100 % of the share capital of the Company, are represented at the present extraordinary general meeting.
- III. All the shareholders declare having been informed in advance on the agenda of the meeting and waived all convening requirements and formalities. The meeting is thus regularly constituted and can validly deliberate and decide on the agenda.
 - IV. The agenda of the Meeting is the following:

Agenda:

To entirely restate the Articles.

The following resolution was unanimously approved by the Shareholders:

Sole resolution

The Shareholders resolve to entirely restate the Articles in the form as follows:

1. Corporate form and Name. This document constitutes the articles of incorporation (the "Articles") of CDR Bounty (Luxembourg) S.à r.l. (the "Company"), a private limited liability company (société à responsabilité limitée) incorporated under the laws of the Grand Duchy of Luxembourg including the law of 10 August 1915 on commercial companies as amended from time to time (the "1915 Law").



2. Registered office.

- 2.1 The registered office of the Company (the "Registered Office") is established in the city of Luxembourg, Grand Duchy of Luxembourg.
 - 2.2 The Registered Office may be transferred:
 - 2.2.1to any other place within the same municipality in the Grand Duchy of Luxembourg by:
 - (a) the Sole Manager (as defined in Article 24.2) if the Company has at the time a Sole Manager; or
 - (b) the Board of Managers (as defined in Article 24.3) if the Company has at the time a Board of Managers; or
- 2.2.2 to any other place in the Grand Duchy of Luxembourg (whether or not in the same municipality) by a resolution of the shareholders of the Company (a "Shareholders' Resolution") passed in accordance with these Articles -including Article 29.4 and the laws of the Grand Duchy of Luxembourg including the 1915 Law as they are in force from time to time ("Luxembourg Law").
- 2.3 Should a situation arise or be deemed imminent, whether military, political, economic, social or otherwise, which would prevent normal activity at the Registered Office, the Registered Office may be temporarily transferred abroad until such time as the situation becomes normalised; such temporary measures will not have any effect on the Company's nationality and the Company will, notwithstanding this temporary transfer of the Registered Office, remain a Luxembourg company. The decision as to the transfer abroad of the Registered Office will be made by the Sole Manager or the Board of Managers as appropriate.
- 2.4 The Company may have branches, subsidiaries or other offices, both in the Grand Duchy of Luxembourg and abroad.

3. Objects. The objects of the Company are:

- 3.1 to act as an investment holding company and to co-ordinate the business of any corporate bodies in which the Company is for the time being directly or indirectly interested, and to acquire (whether by original subscription, tender, purchase, exchange or otherwise) the whole of or any part of the stock, shares, debentures, debenture stocks, bonds and other securities issued or guaranteed by any person and any other asset of any kind and to hold the same as investments, and to sell, exchange and dispose of the same;
- 3.2 to carry on any trade or business whatsoever and to acquire, undertake and carry on the whole or any part of the business, property and/or liabilities of any person carrying on any business;
- 3.3 to invest and deal with the Company's money and funds in any way the Sole Manager or the Board of Managers (as appropriate) thinks fit and to lend money and give credit in each case to any person with or without security;
- 3.4 to borrow, raise and secure the payment of money in any way the Sole Manager or the Board of Managers (as appropriate) thinks fit, including by the issue (to the extent permitted by Luxembourg Law) of debentures and other securities or instruments, perpetual or otherwise, convertible or not, whether or not charged on all or any of the Company's property (present and future) or its uncalled capital, and to purchase, redeem, convert and pay off those securities;
- 3.5 to acquire an interest in, amalgamate, merge, consolidate with and enter into partnership or any arrangement for the sharing of profits, union of interests, co-operation, joint venture, reciprocal concession or otherwise with any person, including any employees of the Company;
- 3.6 to enter into any guarantee or contract of indemnity or suretyship, and to provide security for the performance of the obligations of and/or the payment of any money by any person (including any body corporate in which the Company has a direct or indirect interest or any person (a "Holding Entity") which is for the time being a member of or otherwise has a direct or indirect interest in the Company or any body corporate in which a Holding Entity has a direct or indirect interest and any person who is associated with the Company in any business or venture), with or without the Company receiving any consideration or advantage (whether direct or indirect), and whether by personal covenant or mortgage, charge or lien over all or part of the Company's undertaking, property or assets (present and future) or by other means; for the purposes of this Article 3.6 "guarantee" includes any obligation, however described, to pay, satisfy, provide funds for the payment or satisfaction of, indemnify and keep indemnified against the consequences of default in the payment of, or otherwise be responsible for, any indebtedness or financial obligations of any other person;
- 3.7 to purchase, take on lease, exchange, hire and otherwise acquire any real or personal property and any right or privilege over or in respect of it;
- 3.8 to sell, lease, exchange, let, hire and dispose of any real or personal property and/or the whole or any part of the undertaking of the Company, for such consideration as the Sole Manager or the Board of Managers (as appropriate) thinks fit, including for shares, debentures or other securities, whether fully or partly paid up, of any person, whether or not having objects (altogether or in part) similar to those of the Company; to hold any shares, debentures and other securities so acquired; to improve, manage, develop, sell, exchange, lease, mortgage, dispose of, grant options over, turn to account and otherwise deal with all or any part of the property and rights of the Company;
- 3.9 to do all or any of the things provided in any paragraph of this Article 3 (a) in any part of the world; (b) as principal, agent, contractor, trustee or otherwise; (c) by or through trustees, agents, sub-contractors or otherwise; and (d) alone or with another person or persons;



3.10 to do all things (including entering into, performing and delivering contracts, deeds, agreements and arrangements with or in favour of any person) that are in the opinion of the Sole Manager or the Board of Managers (as appropriate) incidental or conducive to the attainment of all or any of the Company's objects, or the exercise of all or any of its powers;

PROVIDED ALWAYS that the Company will not enter into any transaction which would constitute a regulated activity of the financial sector or require a business license under Luxembourg Law without due authorisation under Luxembourg Law.

4. Duration. The Company is established for an unlimited duration.

5. Share capital.

- 5.1 The share capital of the Company is GBP 1,989,166.73 (one million, nine hundred and eighty-nine thousand, one hundred and sixty-six pounds sterling and seventy-three pence) divided into:
- 5.1.12,166,690 (two million, one hundred and sixty-six thousand, six hundred and ninety) class A ordinary shares (the "Class A Ordinary Shares") itself divided into the following subclasses:
- (1) 433,338 (four hundred and thirty-three thousand, three hundred and thirty-eight) class A1 ordinary shares (the "Class A1 Ordinary Shares");
- (2) 433,338 (four hundred and thirty-three thousand, three hundred and thirty-eight) class A2 ordinary shares (the "Class A2 Ordinary Shares");
- (3) 433,338 (four hundred and thirty-three thousand, three hundred and thirty-eight) class A3 ordinary shares (the "Class A3 Ordinary Shares");
- (4) 433,338 (four hundred and thirty-three thousand, three hundred and thirty-eight) class A4 ordinary shares (the "Class A4 Ordinary Shares"); and
- (5) 433,338 (four hundred and thirty-three thousand, three hundred and thirty-eight) class A5 ordinary shares (the "Class A5 Ordinary Shares");
- 5.1.2 8,699,985 (eight million, six hundred and ninety-nine thousand, nine hundred and eighty-five) class B ordinary shares (the "Class B Ordinary Shares"), itself divided into the following subclasses:
- (1) 1,739,997 (one million, seven hundred and thirty-nine thousand, nine hundred and ninety-seven) class B1 ordinary shares (the "Class B1 Ordinary Shares");
- (2) 1,739,997 (one million, seven hundred and thirty-nine thousand, nine hundred and ninety-seven) class B2 ordinary shares (the "Class B2 Ordinary Shares");
- (3) 1,739,997 (one million, seven hundred and thirty-nine thousand, nine hundred and ninety-seven) class B3 ordinary shares (the "Class B3 Ordinary Shares");
- (4) 1,739,997 (one million, seven hundred and thirty-nine thousand, nine hundred and ninety-seven) class B4 ordinary shares (the "Class B4 Ordinary Shares"); and
- (5) 1,739,997 (one million, seven hundred and thirty-nine thousand, nine hundred and ninety-seven) class B5 ordinary shares (the "Class B5 Ordinary Shares");
- 5.1.3 13,050,000 (thirteen million and fifty thousand) class C ordinary shares (the "Class C Ordinary Shares"), itself divided into the following subclasses:
 - (1) 2,610,000 (two million, six hundred and ten thousand) class C1 ordinary shares (the "Class C1 Ordinary Shares");
 - (2) 2,610,000 (two million, six hundred and ten thousand) class C2 ordinary shares (the "Class C2 Ordinary Shares");
 - (3) 2,610,000 (two million, six hundred and ten thousand) class C3 ordinary shares (the "Class C3 Ordinary Shares");
- (4) 2,610,000 (two million, six hundred and ten thousand) class C4 ordinary shares (the "Class C4 Ordinary Shares"); and
 - (5) 2,610,000 (two million, six hundred and ten thousand) class C5 ordinary shares (the "Class C5 Ordinary Shares");
- 5.1.4174,999,998 (one hundred and seventy four million, nine hundred and ninety-nine thousand, nine hundred and ninety-eight) preference shares (the "Preference Shares");

with a nominal value of one penny (GBP 0.01) each and the rights and obligations set out in these Articles (the "Shares"). Each Shareholder shall hold the same proportion of each Ordinary Share Series as they do of Shares. In these Articles, "Shareholders" means the holders at the relevant time of the Shares and "Shareholder" shall be construed accordingly.

- 5.2 The Company may establish a share premium account (the "Share Premium Account") into which any premium paid on any Share is to be transferred. Any share premium paid is always allocated to the relevant First Series Ordinary Share, Second Series Ordinary Share, Third Series Ordinary Share, Fourth Series Ordinary Share or Fifth Series Ordinary Share of the class of Shares in connection with which it was paid. Decisions as to the use of the Share Premium Account are to be taken by the Manager(s) subject to the 1915 Law and these Articles.
- 5.3 The Company may, without limitation, accept equity or other contributions without issuing Shares or other securities in consideration for the contribution and may credit the contributions to one or more accounts. Decisions as to the use of any such accounts are to be taken by the Manager(s) subject to the 1915 Law and these Articles. For the avoidance of doubt, any such decision may, but need not, allocate any amount contributed to the contributor.
 - 5.4 All Shares have equal rights unless otherwise provided in these Articles.



- 5.5 The share capital of the Company may be redeemed through the repurchase and subsequent cancellation of all the issued shares of one or more class or classes of Shares (a "Share Redemption").
 - 5.6 Rights on issue of further Shares
- 5.6.1 Subject always to the terms of any shareholders' agreement in relation to the Company entered into from time to time and Article 5.7, if from time to time following Completion the Company proposes to issue any Class A Ordinary Shares or Class B Ordinary Shares or Class C Ordinary Shares or other shares (or instruments convertible into shares) in the capital of the Company for cash, other than the Wider Manager Shares, pursuant to Article 5.8 (the "Relevant Securities"), no such Relevant Securities will be so issued unless such issuance is made pursuant to this Article 5.6 and each Ordinary Shareholder has first been given an opportunity which shall remain open for not less than 10 Business Days (such date as chosen being the "End Date") to subscribe, at the same time and on the same terms, for its Relevant Entitlement. Such opportunity shall be offered to each of the Ordinary Shareholders in the form of a notice in writing from the Company and if the Company proposes to offer such Relevant Securities with a corresponding proportion of bonds, PECs, loan notes, preference shares or other securities or debt instruments ("Other Securities"), the notice shall include the relevant terms and conditions of the offer (the "New Issue Notice").
- 5.6.2 For the purposes of this Article 5.6 "Relevant Entitlement" shall mean, in the case of each Ordinary Shareholder, such percentage of the Relevant Securities (with a corresponding proportion of Other Securities) as equates to his or its pro rata share of the Ordinary Shares in issue immediately prior to the issue of the Relevant Securities (save that an Investor's Relevant Entitlement may instead be subscribed by an Affiliate of the Investor and a Manager's Relevant Entitlement may instead be subscribed by a Family Member or Family Trust of the Manager). Each Class A Ordinary Shareholder shall receive his Relevant Entitlement of shares in the form of Class B Ordinary Shares. Each Class C Ordinary Shareholder shall receive his Relevant Entitlement of shares in the form of Class B Ordinary Shares.
- 5.6.3 The New Issue Notice shall indicate the total number of Relevant Securities and Other Securities to be issued, the Relevant Entitlement of each Ordinary Shareholder and the subscription price of each Relevant Security and each Other Security. If an Ordinary Shareholder wishes to subscribe for any or all of its Relevant Entitlement, it shall give notice in writing to the Company on or before the End Date, failing which the Ordinary Shareholder shall be deemed to have declined to subscribe for any or all of its Relevant Entitlement in connection with the New Issue Notice. Any notice given by an Ordinary Shareholder pursuant to this Article 5.6.3 shall be irrevocable.
- 5.6.4 If by 5.00 p.m. on the End Date, the Company has not received notices under Article 5.6.3 in respect of all of the Relevant Securities (the Relevant Securities in respect of which no notice has been received being the "Excess Securities"), the Topco Board shall (with an Investor Consent) offer such Excess Securities to those Ordinary Shareholders who have given notice under Article 5.6.3. Such Ordinary Shareholders shall be given a further reasonable period of time (being not less than 5 Business Days, such date chosen being the "Second End Date") to subscribe for the Excess Securities at the same time pro rata to the number of Ordinary Shares held by such Ordinary Shareholder (save that the Excess Securities may be subscribed by an Affiliate of the Investor in place of the Investor and the Excess Securities may be subscribed by a Family Member or Family Trust in place of the Manager) and on the same terms on which that Ordinary Shareholder subscribed for the Relevant Securities pursuant to the New Issue Notice.
 - 5.6.5 On or before the Second End Date, the Company shall give notice in writing to each Ordinary Shareholder of:
- (a) the number and price of the Relevant Securities (and Excess Securities, as applicable) and the Other Securities for which that Ordinary Shareholder has committed to subscribe; and
- (b) the place and time on which the subscription is to be completed and the account details for the telegraphic transfer of the required subscription monies.
 - 5.7 Emergency Issue and Acquisition Issue
- 5.7.1 If an Emergency Issue becomes necessary in the reasonable opinion of the Topco Board, or an Acquisition Issue is proposed by the Majority Investors and approved with either Majority Shareholder Consent or Majority Board Consent in accordance with the terms of any shareholders' agreement in relation to the Company entered into from time to time, each Shareholder shall:
- (a) consent to any board or shareholders' meeting of a member of the Group being held on short notice to implement the Emergency Issue or Acquisition Issue (as applicable) and to procure (so far as it is able) that any director appointed by it will so consent (subject to his fiduciary duties);
- (b) vote in favour of all resolutions as a shareholder and (subject to his fiduciary duties) as a director of the relevant member of the Group, which are proposed to implement the Emergency Issue or Acquisition Issue (as applicable); and
- (c) procure the circulation to the board of directors or managers as appropriate or shareholders of the relevant member of the Group of such board or shareholder written resolutions (respectively) proposed to implement the Emergency Issue or Acquisition Issue (as applicable) and (subject to their fiduciary duties as a director of the relevant member of the Group) to sign (or to the extent permitted by applicable law in the case of a written resolution, to indicate their agreement to) such resolutions and return them (or the relevant indication) to the Company as soon as possible.
- 5.7.2 Subject to Article 5.7.3, Article 5.6 shall not apply in relation to an Emergency Issue or Acquisition Issue (as applicable) of Ordinary Shares and that, for the purposes of implementing an Emergency Issue or Acquisition Issue (as



applicable), the Majority Investors may, subject to Article 5.8 below, determine the number of Relevant Securities and Other Securities to be issued and the timing and other terms of that issue.

- 5.7.3 Each Manager Party and the Management Trustee shall be entitled (but not obliged) on or before the date which is 45 days immediately after an Emergency Issue or Acquisition Issue (as applicable) of Ordinary Shares, PECs or other securities to subscribe for (or otherwise acquire) such relevant proportion of shares or other securities as he or it would have been entitled to had Article 5.6 applied to such an issue by reference to his holding of Ordinary Shares immediately prior to the Emergency Issue or Acquisition Issue (as applicable), on the same terms as the Investor, provided that the Manager Party and/or Management Trustee as appropriate also acquires his or its pro rata proportion of Other Securities acquired by the Investor as part of the Emergency Issue or Acquisition Issue (as applicable) in the same proportions and on the same terms as the Investor.
- 5.7.4 For the purposes of this Article 5.7, "Acquisition Issue" and "Emergency Issue" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time.
 - 5.8 Wider manager shares

Wider Manager Shares may be issued to and allocated following Completion to such Wider Managers, and on such terms, as may be approved by the Remuneration Committee.

6. Order of payments.

- 6.1 All:
- 6.1.1 net proceeds (whether cash or non-cash) of a Sale, Listing or Option Sale (following repayment of the PECs) ("Sale Payments");
- 6.1.2 payments (whether cash or non-cash) made by the Company pursuant to a Winding Up (following repayment of the PECs); and
- 6.1.3 payments (whether cash or non-cash) made by the Company ("Distributions") by way of dividend ("Dividends") or pursuant to a Share Redemption ("Redemption Payments"),
- to be made to the Shareholders shall on the Ratchet Relevant Date be allocated by the Company among the Shareholders in accordance with Articles 6.2 and 14.
 - 6.2 All Equity Proceeds shall be allocated in the following order and manner:
- 6.2.1 first, an amount equal to the Preference Share Amount to the holders of Preference Shares in accordance with Article 7;
- 6.2.2 second, to the extent of any Equity Proceeds remaining once holders of Preference Shares have received the amounts due to them under Article 6.2.1, an amount equal to the Ordinary Share Amount to the holders of Ordinary Shares in accordance with Article 13;
- 6.2.3 third, to the extent of any Equity Proceeds remaining once holders of Preference Shares and Ordinary Shares have received the amounts due to them under Articles 6.2.1 and 6.2.2, then, and until the amount of the Fund Return is such that the First Target Return has been received by the Investor:
- (a) 13 per cent. of the balance of any Equity Proceeds in respect of that Event (after deduction of the Equity Proceeds in respect of that and all previous Events (if any) allocated under Articles 6.2.1 and 6.2.2) to the holders of Class A Ordinary Shares in accordance with Article 13; and
- (b) 87 per cent. of the balance of any Equity Proceeds in respect of that Event (after deduction of the Equity Proceeds in respect of that and all previous Events (if any) allocated under Articles 6.2.1 and 6.2.2) to the holders of Class B Ordinary Shares and Class C Ordinary Shares in accordance with Article 13;
- 6.2.4 fourth, to the extent of any Equity Proceeds remaining once holders of Preference Shares, Class A Ordinary Shares, Class B Ordinary Shares and Class C Ordinary Shares have received the amounts due to them under Articles 6.2.1, 6.2.2 and 6.2.3 and if the Fund Return is such that the First Target Return has been received by the Investor, then, and until the amount of the Fund Return is such that the Second Target Return has been received by the Investor:
- (a) 15 per cent. of the balance of any Equity Proceeds in respect of that Event (after deduction of the Equity Proceeds in respect of that and all previous Events (if any) allocated under Articles 6.2.1, 6.2.2 and 6.2.3) to the holders of Class A Ordinary Shares in accordance with Article 13; and
- (b) 85 per cent. of the balance of any Equity Proceeds in respect of that Event (after deduction of the Equity Proceeds in respect of that and all previous Events (if any) allocated under Articles 6.2.1, 6.2.2 and 6.2.3) to the holders of Class B Ordinary Shares and Class C Ordinary Shares in accordance with Article 13; and
- 6.2.5 fifth, to the extent of any Equity Proceeds remaining once holders of Preference Shares, Class A Ordinary Shares, Class B Ordinary Shares and Class C Ordinary Shares have received the amounts due to them under Articles 6.2.1, 6.2.2, 6.2.3 and 6.2.4) and if the Fund Return is such that the Second Target Return has been received by the Investor, then, and for the remainder of the Fund Return:
- (a) 20 per cent. of the balance of any Equity Proceeds in respect of that Event (after deduction of the Equity Proceeds in respect of that and all previous Events (if any) allocated under Articles 6.2.1, 6.2.2, 6.2.3 and 6.2.4) to the holders of Class A Ordinary Shares in accordance with Article 13; and



- (b) 80 per cent. of the balance of any Equity Proceeds in respect of that Event (after deduction of the Equity Proceeds in respect of that and all previous Events (if any) allocated under Articles 6.2.1, 6.2.2, 6.2.3 and 6.2.4) to the holders of Class B Ordinary Shares and Class C Ordinary Shares in accordance with Article 13.
- 6.3 If the Event Value is not agreed by the Majority Investors and the Managers' Representative it shall immediately be referred to an independent chartered accountant (the "Ratchet Expert") for determination and certification. The Ratchet Expert shall be an independent chartered accountant of not less than 5 years' standing who shall be nominated by agreement between the Majority Investors and the Managers' Representative (or failing such nomination on or before the date which is 3 days after this expert mechanism being invoked as may be nominated by the president for the time being of the Institute of Chartered Accountants in England and Wales upon the request of any member of the Topco Board). Upon being nominated, the Ratchet Expert shall determine and certify the Event Value, shall act as an expert and not an arbitrator and shall (save in the case of manifest error) be conclusive and binding on the Company and its members. The costs of the Ratchet Expert shall be borne by the Company.
 - 6.4 For the purposes of Articles 6, 13 and 14 the following terms have the meanings set out below:
- 6.4.1 "Asset Sale" has the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;
- 6.4.2 "Available Amount" means the total amount of net profits of the Company for the relevant period in which shares are to be redeemed (including carried forward profits) increased by:
 - (1) any freely distributable reserves attached to the Ordinary Shares; and
- (2) by the amount of the capital reduction and legal reserve reduction relating to the Ordinary Share Series to be cancelled,

but decreased by:

- (1) any losses (including carried forward losses) expressed as a positive value;
- (2) any sums to be placed into reserve(s) pursuant to legal requirements, each time as set out in the relevant interim accounts (without any double counting);
 - (3) any preferred fixed return accumulated on the Preference Shares; and
 - (4) any Ordinary Dividend entitlement,

and must be set out in the interim accounts for the Ordinary Share Series period to be drawn down not earlier than 30 days, but not later than 10 days, before the date of the relevant Share Redemption and shall be determined by the Board of Managers;

- 6.4.3 "Available Liquidities" means:
- (1) all the cash held by the Company (except for cash on term deposits with a remaining maturing exceeding six months);
- (2) any readily marketable money market instruments, bonds and notes, and any receivable which in the opinion of the Board of Managers will be paid to the Company in the short term, less any indebtedness or other debt of the Company payable in less than six months determined on the basis of the interim accounts relating to the relevant series period (or new period, as the case may be); and
 - (3) any assets such as shares, stock or securities of other kind held by the Company;
- 6.4.4 "Deferred Consideration" means any consideration which is deferred including if held in escrow or other contingent or non-contingent rights to receive payment or other consideration in the future (not including Illiquid Securities);
 - 6.4.5 "Equity Proceeds" means the Event Value in respect of each Event;
 - 6.4.6 "Event" means an Exit, completion of an Option Sale or a Distribution, or a combination of some or all of them;
 - 6.4.7 "Event Value" means:
- (a) in the case of a Listing, the price per share at which ordinary shares in the relevant Group Company are sold or offered in connection with the Listing (in the case of an offer for sale, being the underwritten price or, if an offer for sale by tender, the striking price under such offer or, in the case of a placing, the price at which shares are sold under the placing) multiplied by the number of ordinary shares which would be in issue immediately following such Listing (including, for the avoidance of doubt, any shares which are subject to lock-up restrictions), but excluding any ordinary shares issued for the purpose of raising additional or replacement capital for the relevant Group Company as part of the Listing arrangements (whether to refinance the payment of loans or for any other reason whatsoever);
 - (b) in the case of a Sale or an Option Sale the sum of the following:
- (A) cash payable in respect of Shares in connection with the Sale; and (B) the Fair Market Value of all other consideration payable in respect of Shares in connection with the Sale and to the extent a Sale includes a combination of forms of consideration (including contingent consideration), the applicable valuation principle set out in this Article 6.4.7 shall be applied separately, as appropriate, to the relevant element of the consideration and if any element of the consideration is offered in the alternative, the cash alternative, or the most liquid form of security, as applicable, shall be used for the purposes of determining the Event Value;
- (c) in the case of an Asset Sale, the aggregate of all distributions (less any deduction or withholding on account of fees and expenses or tax or as otherwise required by law) to be received by holders of Preference Shares, Class A Ordinary



Shares, Class B Ordinary Shares and Class C Ordinary Shares pursuant to any Distribution following or in connection with the relevant Asset Sale; and

(d) in the case of a Winding-Up or Distribution (other than a Distribution following or in connection with an Asset Sale) the aggregate of all distributions (less any deduction or withholding on account of fees and expenses or tax or as otherwise required by law) to be received by holders of Preference Shares, Class A Ordinary Shares, Class B Ordinary Shares and Class C Ordinary Shares,

in each case, calculated on a fully diluted basis (i.e. save where excluded under Article 6.4.7 (a) and taking account of all shares to be issued on or before the Ratchet Relevant Date) and in any event not exceeding the Available Amount or the Available Liquidities.

For purposes of this Article 6.4.7, "Fair Market Value" of consideration other than immediately available cash means:

- (a) for Marketable Securities, the volume weighted average price for such securities over the 20 trading days prior to the day on which the offer for or intention to acquire the Company is first announced by the proposed purchaser agreed between the Majority Investors and the Managers' Representative or failing agreement as determined by the Ratchet Expert (as defined in Article 6.3);
- (b) for Illiquid Securities, the value of such security agreed between the Majority Investors and the Managers' Representative or failing agreement as determined by the Ratchet Expert, taking into account typical components of value including, but not limited to the liquidity of the securities, and in the case of shares, whether or not they represent a minority interest; and
- (c) for Deferred Consideration, the value of such consideration agreed between the Majority Investors and the Managers' Representative or failing agreement as determined by the Ratchet Expert taking into account typical components of value;
- 6.4.8 "Excluded Fees" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time:
- 6.4.9 "Exit" means completion of a Sale, Listing or Winding-Up or of the Distributions(s) consequent upon an Asset Sale;
- 6.4.10 "First Target Return" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;
- 6.4.11 "Fund Return" means the aggregate amount of all cash received by the Investor from the Company or any third party in respect of any Class C Ordinary Shares, any Preference Shares, any PECs or any other equity, quasi equity or debt securities in the Company held by it and all cash received from time to time prior to the relevant Event by the Investor from any member of the Group or any third party in respect of loans (if any) made by the Investor to the Company or any member of the Group and any loan capital issued by the Company or any member of the Group, including any repayments, conversion or purchases of share capital or PECs, and including any payments made to the Investor pursuant to an Event, (but excluding any Excluded Fees);
- 6.4.12 "Illiquid Securities" means any securities (being shares, loan notes, debentures, bonds or other securities) which are not Marketable Securities;
- 6.4.13 "Marketable Securities" means securities (being shares, loan notes, debentures, bonds or other securities) which are freely tradable without any restriction (other than restrictions agreed to by the relevant holder pursuant to an underwriting agreement, lock-up agreement or otherwise provided they do not exceed 6 months) on any Recognised Investment Exchange, or are otherwise readily saleable within 1 month of the date of receipt;
 - 6.4.14 "Nominated Bank Account" means a client account or other bank account of the Nominated Account Holder;
- 6.4.15 "Nominated Account Holder" means such firm of lawyers or other appropriate entity as is nominated by the Majority Investor and approved by the Managers' Representative such approval not to be unreasonably withheld or delayed;
 - 6.4.16 "Preference Distribution Proportion" of a holder of Preference Shares is the proportion which:
- (a) the total of (i) the nominal value of its Preference Shares and (ii) the accrued but unpaid Preference Dividend on its Preference Shares;

bears to:

- (b) the total of (i) the nominal value of all the Preference Shares in issue at such time and (ii) the accrued but unpaid Preference Dividend on all the Preference Shares in issue at such time;
 - 6.4.17 "Preference Dividend" has the meaning given in Article 8.2;
 - 6.4.18 "Preference Share Amount" means, in respect of any Preference Share:
 - (a) its nominal value; plus
 - (b) any accruals of Preference Dividend;
 - 6.4.19 "Ratchet Relevant Date" means the date of the relevant Event;
- 6.4.20 "Second Target Return" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time; and
 - 6.4.21 "Target Return" means the First Target Return or the Second Target Return (as applicable).



7. Waterfall payments on preference shares.

- 7.1 All Distributions in respect of Preference Shares shall be made by Redemption Payments or by Dividends.
- 7.2 All Equity Proceeds to be allocated in respect of Preference Shares pursuant to Article 6 shall be apportioned between the holders of Preference Shares according to their Preference Distribution Proportions. The Preference Distribution Proportions in respect of any Equity Proceeds shall be calculated as at the date of the Ratchet Relevant Date.

8. Preference share dividends.

- 8.1 Any amount of any Distribution to be distributed to the holders of the Preference Shares by way of Dividend shall be allocated in accordance with Article 6 and this Article 8.
- 8.2 Subject to Article 8.3 below and applicable law, each Preference Share shall be entitled to a fixed dividend ("Preference Dividend") calculated in respect of an Accrual Period at the Applicable Rate on the aggregate of the nominal value of the Preference Share and any Compounded Preference Dividend on the Preference Share and shall accrue daily and be paid in arrears on the Dividend Payment Date in respect of the relevant Accrual Period.
- 8.3 As at any date of determination, any Preference Dividend in respect of a prior Accrual Period which is not paid on the applicable Dividend Payment Date in respect of that Accrual Period, to the extent that it has not subsequently been repaid, will be compounded as envisaged by Article 8.2 ("Compounded Preference Dividend").
 - 8.4 For the purposes of this Article 8:
- 8.4.1 "Accrual Period" means, in relation to any Preference Share, (i) in relation to the initial Accrual Period, the period commencing on and including the Completion Date and ending on (but excluding) the 12 monthly anniversary thereof and (ii) in relation to each subsequent Accrual Period, the 12 month period commencing immediately after the end of the previous Accrual Period. Provided that, if any Preference Share is redeemed during any such twelve month period, that Accrual Period shall end on the Ratchet Relevant Date (as defined in Article 6);
- 8.4.2 "Applicable Rate" means a rate equal to 12 per cent. per annum, computed on a 365/366-day year and the actual number of days elapsed; and
- 8.4.3 "Dividend Payment Date" means the last day of an Accrual Period or if that day is not a Business Day, the next following Business Day.

9. Ordinary share dividends.

- 9.1 Any amount of any Distribution to be distributed to the holders of the Ordinary Shares by way of Dividend shall be allocated in accordance with Article 6 and this Article 9.
- 9.2 Subject to Article 9.3 below and applicable law, each Ordinary Share shall be entitled to a fixed dividend ("Ordinary Dividend") calculated in respect of an Ordinary Accrual Period at:
 - 9.2.1 in respect of the First Series Ordinary Shares, 0.05% per annum;
 - 9.2.2 in respect of the Second Series Ordinary Shares, 0.10% per annum;
 - 9.2.3 in respect of the Third Series Ordinary Shares, 0.15% per annum;
 - 9.2.4 in respect of the Fourth Series Ordinary Shares, 0.20% per annum; and
 - 9.2.5 in respect of the Fifth Series Ordinary Shares, 0.25% per annum,
- on the aggregate of the nominal value of such Ordinary Share and any Compounded Ordinary Dividend on the Ordinary Share and shall accrue daily and be paid in arrears on the Ordinary Dividend Payment Date in respect of the relevant Ordinary Accrual Period.
- 9.3 As at any date of determination, any Ordinary Dividend in respect of a prior Ordinary Accrual Period which is not paid on the applicable Ordinary Dividend Payment Date in respect of that Ordinary Accrual Period, to the extent that it has not subsequently been repaid, will be compounded as envisaged by Article 9.2 ("Compounded Ordinary Dividend").
- 9.4 Subject to applicable law and to the terms of any shareholders' agreement in relation to the Company entered into from time to time, holders of Ordinary Shares may receive a final dividend by resolution of the Shareholders of the Company or an interim dividend if the Board of Managers so determines.
 - 9.5 For the purposes of this Article 9:
- 9.5.1 "Ordinary Accrual Period" means, in relation to any Ordinary Share, (i) in relation to the initial Ordinary Accrual Period, the period commencing on and including the Completion Date and ending on (but excluding) December 31, 2014 and (ii) in relation to each subsequent Ordinary Accrual Period, the 12 month period commencing immediately after the end of the previous Ordinary Accrual Period. Provided that, if any Ordinary Share is redeemed during any such twelve month period, that Ordinary Accrual Period shall end on the Ratchet Relevant Date (as defined in Article 6); and
- 9.5.2 "Ordinary Dividend Payment Date" means the last day of an Accrual Period or if that day is not a Business Day, the next following Business Day.
- **10.** Redemption payments, Winding up payments and Sale payments on preference shares. Each Preference Share carries a right to payment of the Preference Share Amount on a Sale Payment, Winding Up or Redemption Payment.
 - 11. Ranking of preference shares. All Preference Shares shall rank pari passu.



12. Redemption payments, Winding up payments and Sale payments on ordinary shares. Each Ordinary Share carries a right to payment of the Ordinary Share Amount on a Sale Payment, Winding Up or Redemption Payment.

13. Waterfall payments on ordinary shares.

- 13.1 All Distributions in respect of Ordinary Shares shall be made by Redemption Payments or by Dividends.
- 13.2 All Equity Proceeds shall be allocated in respect of Ordinary Shares in accordance with Article 6.2 and:
- (a) in the case of holders of Class A Ordinary Shares, shall be apportioned between such holders pro rata to their holdings of the relevant class of Class A Ordinary Shares;
- (b) in the case of holders of Class B Ordinary Shares, shall be apportioned between such holders pro rata to their holdings of the relevant class of Class B Ordinary Shares; and
- (c) in the case of holders of Class C Ordinary Shares, shall be apportioned between such holders pro rata to their holdings of the relevant class of Class C Ordinary Shares.
- 13.3 Any Share Redemption in respect of Ordinary Shares shall be made in the following order and within the following periods:
 - (a) first, the Class A1, Class B1 and Class C1 Ordinary Shares between January 1, 2013 and December 31, 2014;
 - (b) second, the Class A2, Class B2 and Class C2 Ordinary Shares between January 1, 2015 and December 31, 2015;
 - (c) third, the Class A3, Class B3 and Class C3 Ordinary Shares between January 1, 2016 and December 31, 2016;
 - (d) fourth, the Class A4, Class B4 and Class C4 Ordinary Shares between January 1, 2017 and December 31, 2017;
- (e) fifth, the Class A5, Class B5 and Class C5 Ordinary Shares between January 1, 2018 and December 31, 2018; and thereafter, in case any class of Ordinary Shares is not redeemed in its relevant redemption period, every fifth anniversary of the initial redemption period in respect of that class of Ordinary Shares (the "Subsequent Redemption Period"), provided that if a class of Ordinary Shares has been redeemed (i) the starting date of the redemption period of the next following class of Ordinary Shares to be redeemed shall be the first Business Day following the date of redemption of such redeemed Ordinary Shares but the closing date of that redemption period remains 31 December of the following year and otherwise (ii) each other Subsequent Redemption Period for all remaining classes of Ordinary Shares shall be brought forward by one year per redeemed class of Ordinary Shares.

14. Mechanics on an event.

- 14.1 All Equity Proceeds in the form of cash ("Cash Equity Proceeds") will be paid to the Nominated Bank Account before being distributed in accordance with Article 6.
- 14.2 Subject to Article 14.6, the Board will prepare an apportionment of the Cash Equity Proceeds among the Shareholders in accordance with Articles 6, 7 and 13 (the "Equity Proceeds Apportionment") and will instruct the Nominated Account Holder to distribute the Cash Equity Proceeds to the Shareholders in accordance with the Equity Proceeds Apportionment.
- 14.3 The Nominated Account Holder will not have any liability to any Shareholder for distributing the Cash Equity Proceeds in accordance with the Equity Proceeds Apportionment and each Shareholder including each Majority Investor will enter into such confirmation and/or waiver agreements with the Nominated Account Holder as are reasonably approved by the Majority Investors in relation to the payment and distribution of the Cash Equity Proceeds.
- 14.4 Subject to Article 14.6, all Equity Proceeds in a form other than cash ("Non-cash Equity Proceeds") before being distributed in accordance with Article 6 will be apportioned by the Company, or in respect of a Sale Payment that the relevant payer of the Non-cash Equity Proceeds shall apportion such proceeds, in accordance with the Equity Proceeds Apportionment.
- 14.5 The relevant person apportioning the Non-cash Equity Proceeds pursuant to Article 14.4 will not have any liability to any Shareholder for distributing the Non-cash Equity Proceeds in accordance with the Equity Proceeds Apportionment and each Shareholder, including each Majority Investor, will enter into such confirmation, waiver, assignment, transfer and/or novation agreements with such persons as are reasonably approved by the Majority Investors in relation to the distribution of the Non-cash Equity Proceeds.
- 14.6 Subject always to the terms of any shareholders' agreement in relation to the Company entered into from time to time, if on the Ratchet Relevant Date in respect of an Exit, the aggregate amount paid to and including that date by the Company pursuant to the Manager PEC Instrument in respect of the Cash Yield on all Manager PECs in issue from time to time is:
- 14.6.1 greater than the Threshold (the "Cash Yield Excess"), then, prior to any apportionment which takes place pursuant to Articles 6, 7 and 13, an amount of the Equity Proceeds which is equal to the Cash Yield Excess shall be apportioned and distributed to the Class C Ordinary Shareholder(s) in accordance with Article 13; or
- 14.6.2 less than the Threshold (the "Cash Yield Shortfall"), then, prior to any apportionment which takes place pursuant to Articles 6, 7 and 13, an amount of the Equity Proceeds which is equal to the Cash Yield Shortfall shall be apportioned and distributed to the Class B Ordinary Shareholder(s) in accordance with Article 13;
- and any amounts so apportioned shall be excluded from all calculations under this Article 14. For the purposes of this Article 14.6, the "Threshold" shall mean the amount derived by the following formula:



£22,666,660 \times (A/3,650),

where "A" equals the number of calendar days between and including the Completion Date and the Ratchet Relevant Date

- 14.7 Notwithstanding any other provision of these Articles to the contrary, no Manager Party shall be required to accept Non-cash Equity Proceeds in the form of any shares or other securities in a company incorporated in a jurisdiction different to the jurisdiction of incorporation of the Company on an Exit, an Option Sale, Distributions or Dividends.
- **15. Indivisibility of shares.** Towards the Company, the Company's Shares are indivisible since only one owner is admitted per Share. Joint co-owners have to appoint a sole person as their representative towards the Company.

16. Transfer of shares.

- 16.1 During such time as the Company has only one Shareholder, the Shares will be freely transferable.
- 16.2 Notwithstanding any provision to the contrary in these Articles, the Company shall not register a transfer of Shares unless the transfer is made in accordance with article 189 and 190 of the 1915 Law and the transfer is permitted by Article 17.
- 16.3 An Ordinary Shareholder is not entitled to Dispose of Ordinary Shares or Shareholder Instruments unless the Disposal is permitted by Article 17.
- 16.4 To the extent that a transfer complies with Article 16.2, the Board of Managers shall promptly register the transferee in the register of shareholders or Investor PEC holders or Manager PEC holders as the case may be of the Company.
- 16.5 The Board of Managers (with an Investor Consent) may (and shall if so required by an Investor Director) require such information or evidence as is reasonable to ensure that a relevant Disposal is permitted under the Articles and any shareholders' agreement in relation to the Company entered into from time to time.

17. Transfer restrictions.

- 17.1 Other than pursuant to the Articles 17.2.3, 17.2.4, 17.2.5, 17.3.2 or 17.3.7 but otherwise notwithstanding any provision to the contrary in these Articles, no Ordinary Share or Shareholder Instrument may be Disposed of before the date which is twenty-four months following Completion (the "Lock-Up Period") without Majority Shareholder Consent.
- 17.2 Subject to Articles 17.1 and 17.6, no Ordinary Share or Shareholder Instrument may be Disposed of by the Managers except:
 - 17.2.1 with Investor Consent (which consent may be granted unconditionally or subject to terms or conditions);
- 17.2.2 following the date which is six years following Completion, pursuant to the right of first offer procedures set out in Article 18;
 - 17.2.3 when the Disposal is required by, and made pursuant to, Article 22;
 - 17.2.4 to a Family Member or Family Trust;
- 17.2.5 other than in respect of Class A Ordinary Shares for the twelve months immediately following the relevant Cessation Date (during which period Article 22 will apply), notwithstanding Article 17.1 and subject to the right of first offer procedures set out in Article 18, by the estate of a Manager (including, where the Manager is a trust, its beneficiaries) to a third party where that Manager has died;
 - 17.2.6 in acceptance of a Tag Offer made in accordance with Article 20;
 - 17.2.7 when required by Article 19;
 - 17.2.8 pursuant to a Listing (in accordance with the provisions of any underwriting agreement entered into); or
 - 17.2.9 pursuant to an Option Sale in accordance with Article 23.
 - 17.3 Subject to Article 17.1, no Ordinary Share or Shareholder Instrument may be Disposed of by the Investor except:
 - 17.3.1 with the consent of the Managers' Representative;
 - 17.3.2 to an Affiliate of the Investor;
 - 17.3.3 pursuant to the right of first offer procedures set out in Article 18;
 - 17.3.4 pursuant to a proposed Tag Sale which gives rise to a requirement under Article 20 for a Tag Offer;
 - 17.3.5 pursuant to a proposed Sale which gives rise to a right to serve a Drag Notice in accordance with Article 19;
 - 17.3.6 pursuant to a Listing (in accordance with the provisions of any underwriting agreement entered into);
 - 17.3.7 to a Syndicatee; or
 - 17.3.8 pursuant to an Option Sale in accordance with Article 23.
- 17.4 If a Disposal is made to a Family Member who thereafter ceases to be a Family Member (whether by divorce or otherwise) or to a Family Trust which thereafter ceases to be a Family Trust, such Related Holder, as applicable, shall promptly notify the Investor Directors in writing and the Board of Managers may (and shall, if requested by the Investor Directors) authorise any member of the Board of Managers to execute, complete and deliver as agent for and on behalf of that Related Holder a transfer of all of the Class A Ordinary Shares, Class B Ordinary Shares and Shareholder Instruments then held by that Related Holder to the relevant employee, director or consultant to whom they were originally



allocated or to a Related Holder of that employee, director or consultant for nominal consideration, which transfer shall constitute an implied warranty from the relevant Related Holder in favour of the relevant transferee that the legal and beneficial title to the relevant Class A Ordinary Shares, Class B Ordinary Shares and Shareholder Instruments was transferred to the relevant transferee free from all Encumbrances and with full title guarantee. Subject to due stamping (if applicable), the Board of Managers shall register such transfer, after which the validity of such transfer shall not be questioned by any person.

17.5 If a Disposal is made to an Affiliate who thereafter ceases to be an Affiliate (a "Disqualified Affiliate"), the Disqualified Affiliate shall promptly transfer the Ordinary Shares and Shareholder Instruments then held by that Disqualified Affiliate to the relevant Investor to whom they were originally allocated or to an Affiliate of the Investor for nominal consideration, and the Board of Managers may (and shall if requested by the Investor Directors) authorise any member of the Topco Board to execute, complete and deliver as agent for and on behalf of that Disqualified Affiliate a transfer of the Ordinary Shares and Shareholder Instruments, which transfer shall constitute an implied warranty from the Disqualified Affiliate in favour of the relevant transferee that the legal and beneficial title to the relevant Ordinary Shares and Shareholder Instruments was transferred to the relevant transferee free from all Encumbrances and with full title guarantee. Subject to due stamping (if applicable), the Board of Managers shall register such transfer, after which the validity of such transfer shall not be questioned by any person.

17.6 Other than with Majority Shareholder Consent, following any Disposal of Class A Ordinary Shares or Class B Ordinary Shares in accordance with Article 17.2.5 above the recipient of such Disposal shall not:

17.6.1 be entitled to appoint any director or manager of, or observer to, any Group Company pursuant to Article 24;

17.6.2 be entitled to be supplied with any information or documents to be supplied by any party pursuant to any shareholders' agreement in relation to the Company entered into from time to time;

17.6.3 have any right to provide (alone or in conjunction with any other person) any consent required under Article 30: or

17.6.4 be entitled to serve notice on any person to pursue an Exit or Refinancing pursuant to any shareholders' agreement in relation to the Company entered into from time to time.

17.7 In the event that any Class A Ordinary Shareholder or Class B Ordinary Shareholder Disposes of some or all of its Class A Ordinary Shares or Class B Ordinary Shares to a Class C Ordinary Shareholder in accordance with this Article 17, each such disposing Shareholder shall give notice in writing of such Disposal to the Company (copied to Topco) and the general meeting of the Shareholders of the Company shall promptly redesignate such Class A Ordinary Shares or Class B Ordinary Shares as Class C Ordinary Shares.

17.8 In the event that any Class C Ordinary Shareholder Disposes of some or all of its Class C Ordinary Shares to a Class A Ordinary Shareholder or a Class B Ordinary Shareholder in accordance with this Article 17, each such disposing Shareholder shall give notice in writing of such Disposal to the Company (copied to Topco) and the general meeting of the Shareholders of the Company shall promptly redesignate such Class C Ordinary Shares as Class A Ordinary Shares or Class B Ordinary Shares (as appropriate) (or, in the event that the recipient of the relevant Class C Ordinary Shares holds both Class A Ordinary Shares and Class B Ordinary Shares, into Class A Ordinary Shares and Class B Ordinary Shares in proportion to the recipient's existing holding of Class A Ordinary Shares and Class B Ordinary Shares).

17.9 Any Disposal or purported Disposal in breach of Articles 17.1, 17.2, 17.3 shall be void and shall have no effect and the provisions of Article 17.1 shall apply to it.

18. Right of first offer.

18.1 This Article 18 applies if one or more Class B Ordinary Shareholders or Class C Ordinary Shareholders (the "Proposed Transferor(s)") propose (whether through a single transaction or a series of related transactions) to transfer (including pursuant to an Exit) the legal and beneficial title to any Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable), other than pursuant to Articles 17.2.1, 17.2.3, 17.2.4, 17.2.8, 17.2.9, 17.3.1, 17.3.2, 17.3.6, 17.3.7 and 17.3.8 (a "Pre-Emption Transfer").

18.2 Prior to entering into a binding agreement in relation to a Pre-Emption Transfer, the Proposed Transferor(s) must give notice in writing (a "Pre-Emption Notice") to the Company (as agent for and on behalf of the Class B Ordinary Shareholders and Class C Ordinary Shareholders other than the Proposed Transferor(s)) (each an "Eligible Ordinary Shareholder") setting out the number of Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable) (the "Pre-Emption Shares") that the Proposed Transferor(s) wish to transfer pursuant to the Pre-Emption Transfer.

18.3 By the fifth Business Day following the date of receipt by the Company of a Pre-Emption Notice, the Company shall send each Eligible Ordinary Shareholder a copy of the Pre-Emption Notice, together with a statement of its Pre-Emption Proportion of the Pre-Emption Shares.

18.4 By the thirtieth day following the date of the Pre-Emption Notice (or such other date and time as is agreed between the Proposed Transferor(s) and the Eligible Ordinary Shareholder(s)) (the "Pre-Emption Offer Closing Date"), any Eligible Ordinary Shareholder (a "Pre-Emption Offeror") may by notice in writing to the Company and copied to the Proposed Transferor(s) (the "Pre-Emption Offer") offer to acquire:

18.4.1 some or all of its Pre-Emption Proportion of the Pre-Emption Shares; and

18.4.2 some or all of the Pre-Emption Shares:



- (a) for which Pre-Emption Offers are not made by other Eligible Ordinary Shareholders; and/or
- (b) in relation to which Pre-Emption Offers made by other Eligible Ordinary Shareholders are rejected by the Proposed Transferor(s),

(together, the "Excess Pre-Emption Shares").

- 18.5 A Pre-Emption Offer must set out the cash price per relevant Pre-Emption Share (the "Pre-Emption Price") and any other terms on which the relevant Pre-Emption Offeror offers to acquire the relevant Pre-Emption Shares. Once made, a Pre-Emption Offer shall be irrevocable and binding and shall be accepted or rejected by the Proposed Transferor (s) in accordance with Article 18.8.
- 18.6 If an Eligible Ordinary Shareholder fails to submit a Pre-Emption Offer by the Pre-Emption Offer Closing Date, such Eligible Ordinary Shareholder shall be deemed to have declined to make a Pre-Emption Offer and shall have no further rights under this Article 18 in relation to the Pre-Emption Shares.
- 18.7 If Pre-Emption Offers are received for a number of Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable) in excess of the total number of Pre-Emption Shares, each Pre-Emption Offeror who offered to buy Excess Pre-Emption Shares in accordance with Article 18.4.2 shall be deemed for all purposes to have offered to buy its Excess Pre-Emption Shares Proportion of the Excess Pre-Emption Shares.
- 18.8 By the fifth Business Day following the Pre-Emption Offer Closing Date, the Proposed Transferor(s) must inform the Company in writing whether they accept or reject each Pre-Emption Offer. As soon as reasonably practicable thereafter, the Company shall:
- 18.8.1 give notice in writing to each Pre-Emption Offeror whose Pre-Emption Offer has been rejected of that fact (a "Rejection Notice"); and
- 18.8.2 give notice in writing (an "Acceptance Notice") to each Pre-Emption Offeror whose Pre-Emption Offer has been accepted (an "Accepted Offeror") of the number of Pre-Emption Shares it is obliged to acquire, calculated in accordance with Articles 18.4 and 18.7.
 - 18.9 Each Acceptance Notice shall state:
- 18.9.1 a date, place and time (the "Pre-Emption Completion Date") between 10 and 15 Business Days after the Pre-Emption Offer Closing Date (or such other date, place and time as the Proposed Transferor(s) and the Accepted Offeror (s) may agree), on which the sale and purchase of the relevant Pre-Emption Shares is to be completed; and
- 18.9.2 the Proposed Transferor(s) from whom the Accepted Offeror shall acquire the relevant Pre-Emption Shares (failing which, this shall be apportioned pro rata to the number of Pre-Emption Shares to be transferred by the Proposed Transferor(s)).
- 18.10 On or before the Pre-Emption Completion Date, the Proposed Transferor(s) shall transfer the legal and beneficial title to the relevant Pre-Emption Shares to the relevant Accepted Offeror with full title guarantee and free from all Encumbrances by delivering to the Company:
- 18.10.1 if a certificate has been issued, the relevant share certificate(s) (or an indemnity in respect thereof in a form satisfactory to the Topco Board); and
- 18.10.2 a duly executed sale agreement in a form agreed by the Proposed Transferor under which the Proposed Transferor will provide warranties only with respect to its title to, and ownership of, the relevant Pre-Emption Shares and will transfer on the Pre-Emption Completion Date the legal and beneficial title to its relevant Pre-Emption Shares to the relevant Accepted Offeror free from all Encumbrances and with full title guarantee, against payment of the aggregate Pre-Emption Price due to it/them from the relevant Accepted Offeror on the Pre-Emption Completion Date.
- 18.11 The Company's receipt of the aggregate Pre-Emption Price due from each Accepted Offeror in respect of the relevant Pre-Emption Shares shall be a good discharge to the relevant Accepted Offeror. Upon satisfaction of the relevant Proposed Transferor's obligations under Article 18.10 and receipt by the Company of the aggregate Pre-Emption Price due from each Accepted Offeror in respect of the relevant Pre-Emption Shares, the Board of Managers shall register the relevant transfer subject to due stamping (if required).
- 18.12 If by the Pre-Emption Completion Date, an Accepted Offeror fails to pay (or procure the payment of) the aggregate Pre-Emption Price in respect of the relevant Pre-Emption Shares pursuant to Article 18.10 (a "Defaulting Accepted Offeror"), the Proposed Transferor(s) shall (without prejudice to any other rights which it/they may have against that Defaulting Accepted Offeror) be entitled to transfer the legal and beneficial title to such Pre-Emption Shares in accordance with Article 18.13, as if the relevant Eligible Ordinary Shareholder had failed to submit a Pre-Emption Offer in relation to the relevant Pre-Emption Shares, and the Defaulting Accepted Offeror shall have no claim for damages or compensation (or otherwise) against the Proposed Transferor(s) in respect of such Pre-Emption Shares.
- 18.13 The Proposed Transferor(s) may, on or before the date which is 6 months after the Pre-Emption Offer Closing Date, transfer the legal and beneficial title to:
- 18.13.1 those Pre-Emption Shares for which Pre-Emption Offers were not received by the Pre-Emption Offer Closing Date; and
- 18.13.2 those Pre-Emption Shares for which Pre-Emption Offers are deemed not to have been received under Article 18.1218.12, to any person and on any terms, without following the pre-emption procedure set out in this Article 18.



18.14 If the Proposed Transferor(s) reject any Pre-Emption Offer in accordance with Article 18.8.1, the Proposed Transferor(s) may, on or before the date which is 90 days after the relevant Pre-Emption Offer Closing Date, enter into a duly executed sale agreement to transfer the legal and beneficial title to some or all of the Pre-Emption Shares to which the Rejection Notice relates, to any person(s) and on any terms, save that the transfer shall be at no less than the Pre-Emption Price for the relevant Pre-Emption Shares (and the value at such date of any deferred or contingent consideration forming part of the Preemption Price shall be assessed by the Proposed Transferor(s) acting reasonably).

18.15 Whilst the relevant Pre-Emption Shares are the subject of a Pre-Emption Offer, such Pre-Emption Shares may not be Disposed of otherwise than in accordance with this Article 18 without the consent of the Investor or Managers' Representative (as applicable).

18.16 If the Proposed Transferor holds any Shareholder Instruments, the Pre-Emption Notice shall set out such proportion of the Shareholder Instruments held by the Proposed Transferor as the proportion of Pre-Emption Shares bears to the total number of Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable) held by the Proposed Transferor(s) prior to the Pre-Emption Transfer (the "Pre-Emption Shareholder Instruments"). The relevant provisions of this Article 18 shall apply to the Pre-Emption Shareholder Instruments and references to "Pre-Emption Shares" shall be construed accordingly.

19. Drag along rights.

19.1 If, following the end of the Lock-Up Period, a Sale is proposed, any member of the Proposed Buyer Group or any Proposed Seller(s) may, following execution of a binding agreement (whether conditional or unconditional) for the sale of Class C Ordinary Shares to a member of the Proposed Buyer Group (the "Sale Agreement"), by serving a notice in writing (a "Drag Notice") on each holder of Class A Ordinary Shares and/or Class B Ordinary Shares who is not a party to the Sale Agreement (each a "Dragged Seller"), require that Dragged Seller to transfer the same proportion of the Class A Ordinary Shares and/or Class B Ordinary Shares (as applicable) registered in its name (the "Drag Shares") as the proportion of Class C Ordinary Shares to be transferred by the Proposed Seller(s) bears to the total number of Class C Ordinary Shares held by the Proposed Seller(s) prior to the transfer (and if this leads to a fractional entitlement, the number of Class A Ordinary Shares and/or Class B Ordinary Shares (as applicable) should be rounded up to the nearest whole number) (the "Relevant Proportion") to one or more persons identified in the Drag Notice (each a "Drag Buyer") at the consideration indicated in Article 19.2 (the "Drag Price") on the date indicated in the Drag Notice (the "Drag Completion Date"), being not less than 10 Business Days after the date of the Drag Notice and not prior to the date of completion of the Sale Agreement, and on the terms set out in this Article 19. If the Sale Agreement does not complete, the Drag Notice shall lapse and the provisions of this Article 19 shall cease to apply in relation to that Drag Notice.

- 19.2 The consideration for each Drag Share shall:
- 19.2.1 subject to Article 19.2.2, be equal to the higher of:
- (a) the highest consideration offered for each Class C Ordinary Share in the Sale Agreement; and
- (b) the highest consideration paid for any Class C Ordinary Share by the Drag Buyer (or any party with whom the Drag Buyer is acting in concert) in the 6 months prior to the date of the Drag Notice;
- 19.2.2 notwithstanding Article 19.2.1, if the exercise of rights under Article 19.1 takes place after the process set out in Article 18 has been followed then the consideration for each Drag Share shall be equal to the average price per share for the Class C Ordinary Shares to be sold under the Sale Agreement; and
- 19.2.3 subject to Articles 19.3 and 19.8, be in the same form as that offered for each Class C Ordinary Share in the Sale Agreement, shall be paid at the same time as the consideration is payable under the Sale Agreement (or, if later, on the Drag Completion Date) and shall be subject to the same payment terms.
- 19.3 For the purposes of Articles 19.1 and 19.2 "consideration" shall (unless the Majority Investors and the Majority Managers agree otherwise):
- 19.3.1 be satisfied in cash or in freely traded securities listed on one of the Recognised Investment Exchanges located in London, Paris, Frankfurt, Hong Kong, Tokyo, New York or Luxembourg;
- 19.3.2 exclude any offer to subscribe for or acquire any share, debt instrument or other security in the capital of any member of the Proposed Buyer Group made to an Ordinary Shareholder if such offer is an alternative (whether in whole or in part) to the consideration offered for each Class C Ordinary Share under the terms of the Sale Agreement; and
- 19.3.3 for the avoidance of doubt, exclude any right offered to an Ordinary Shareholder to subscribe for or acquire any share, debt instrument or other security in the capital of any member of the Proposed Buyer Group that is separate to the consideration offered for each Class C Ordinary Share under the terms of the Sale Agreement.
- 19.4 Each Dragged Seller shall pay its pro rata share (as a deduction from the gross pre tax proceeds to be received, without prejudice to any other deductions lawfully required to be made) of the third party costs incurred by the Proposed Seller(s) in connection with the proposed Sale and the transfer of the Drag Shares, to the extent that such costs have been incurred on behalf of the Proposed Seller(s) and all of the Dragged Sellers.
- 19.5 Each Dragged Seller shall transfer the legal and beneficial title to its Drag Shares to the Drag Buyer(s) on the terms set out in this Article 19, by delivering to the Company on behalf of the Drag Buyer(s) on or before the Drag Completion Date:



- 19.5.1 if a certificate has been issued, the relevant share certificate(s) (or an indemnity in respect thereof in a form satisfactory to the Topco Board); and
- 19.5.2 a duly executed sale agreement or form of acceptance in a form agreed by the Proposed Seller(s) under which the Dragged Seller will only provide warranties with respect to its title to, and ownership of, the Drag Shares and will transfer on the Drag Completion Date the legal and beneficial title to its Drag Shares to the Drag Buyer free from all Encumbrances and with full title guarantee,

and, to the extent required by the Proposed Seller(s), shall sign such other documents provided that the terms and conditions that apply to the Proposed Seller(s) are not better than those applying to each Dragged Seller in such other documents (and do not require the Dragged Sellers to provide any representation, warranty or undertaking other than pursuant to Article 19.5.2 above) to effect the issue of any shares, debt instruments or other securities to the Dragged Seller.

19.6 The Proposed Buyer Group will pay to the Company the aggregate Drag Price due in respect of all of the Drag Shares on or prior to the Drag Completion Date. Thereafter, the Company shall release the aggregate Drag Price due to each Dragged Seller under Article 19.1 in respect of its Drag Shares following delivery to the Company by that Dragged Seller of the documents required under Article 19.5.

19.7 If a Dragged Seller fails to comply with its obligations under Article 19.5 above (a "Defaulting Dragged Seller"), any member of the Topco Board is authorised to execute, complete and deliver as agent for and on behalf of that Dragged Seller each of the documents referred to in Article 19.5. Subject to due stamping (if any), the Board of Managers shall register the transfer(s), after which the validity of such transfer(s) shall not be questioned by any person. If, under Article 19.3.1 and for the purposes of Articles 19.2 and 19.13, the "consideration" includes an offer to subscribe for or acquire any share, debt instrument or other security in the capital of any member of the Proposed Buyer Group as an alternative (whether in whole or in part), the director so authorised shall have full and unfettered discretion to elect which alternative to accept in respect of each Defaulting Dragged Seller (and may elect for different alternatives for different Defaulting Dragged Sellers) and neither the Topco Board, nor the Board of Managers, nor the director so authorised shall have any liability to such Defaulting Dragged Sellers in relation thereto.

19.8 If a certificate has been issued, each Defaulting Dragged Seller shall surrender its share certificate(s) relating to its Drag Shares (or provide an indemnity in respect thereof in a form satisfactory to the Topco Board) to the Company. On, but not before, such surrender or provision, the Defaulting Dragged Seller shall be entitled to the aggregate Drag Price for its Drag Shares transferred on its behalf without interest and to a new share certificate relating to its remaining holding of Ordinary Shares. Payment to the Dragged Seller(s) shall be made in such manner as is agreed between the Company and the Dragged Seller(s) and in the absence of such agreement, by cheque to the relevant Dragged Seller's last known address. Receipt of the aggregate Drag Price for the Drag Shares so transferred shall constitute an implied warranty from the relevant Dragged Seller(s) in favour of the Drag Buyer(s) that the legal and beneficial title to the relevant Drag Shares was transferred free from all Encumbrances and with full title guarantee.

19.9 The Ordinary Shareholders acknowledge and agree that the authority conferred under Article 19.7 is necessary as security for the performance by the Dragged Seller(s) of their obligations under this Article 19.

19.10 Subject to Article 19.11, unless the Majority Investors or the relevant member of the Proposed Buyer Group otherwise agree in writing, the Dragged Sellers undertake not to attend or vote at any general meeting of the Company or (subject to the 1915 Law) at any meeting of the holders of any class of shares in the capital of the Company or for the purposes of a written resolution of the Company with automatic effect from the date of the Drag Notice (or the date of acquisition of such shares, if later), and consent to not receiving notice of any such meeting and to its Drag Shares not being counted in determining the total number of votes which may be cast at any such meeting or required for the purposes of a written resolution or for the purposes of any other consent required under the Articles.

19.11 The rights referred to in Article 19.10 shall be restored immediately upon the Company registering a transfer of the Drag Shares in accordance with this Article 19.

19.12 If any shares of any class are issued by the Company at any time after the date of the Drag Notice(s) (whether as a result of an Ordinary Shareholders' shareholding or by virtue of the exercise of any right or option or otherwise) (the "Subsequent Shares"), the Proposed Buyer Group shall be entitled to serve an additional notice (a "Further Drag Notice") on each holder of such shares (a "Further Dragged Seller") requiring them to transfer their Relevant Proportion of the Subsequent Shares (the "Subsequent Dragged Shares") to one or more persons identified in the Further Drag Notice at the consideration indicated in Article 19.2 on the date indicated in the Further Drag Notice(s) (the "Further Drag Completion Date"). The provisions of this Article 19 shall apply to the Subsequent Dragged Shares, with the following amendments:

- 19.12.1 references to the "Drag Notice(s)" shall be deemed to be references to the "Further Drag Notice(s)";
- 19.12.2 references to the "Drag Share(s)" shall be deemed to be references to the "Subsequent Dragged Share(s)";
- 19.12.3 references to the "Drag Completion Date" shall be deemed to be references to the "Further Drag Completion Date"; and
 - 19.12.4 references to a "Dragged Seller" shall be deemed to be references to a "Further Dragged Seller".



- 19.13 If the Drag Buyer has also agreed to purchase any Shareholder Instruments from the Proposed Seller(s) in the Sale Agreement, to the extent that some or all of the Dragged Sellers also hold those Shareholder Instruments, the Drag Notice may also require each of the Dragged Sellers to transfer the same proportion of the relevant Shareholder Instruments held by it (the "Drag Shareholder Instruments") as the proportion of the relevant Shareholder Instruments to be transferred by the Proposed Seller(s) bears to the total number of the relevant Shareholder Instruments held by the Proposed Seller(s) prior to the transfer, to the Proposed Buyer Group on the Drag Completion Date at such consideration as is equal to the highest consideration offered for the relevant Shareholder Instruments by the Drag Buyer in the Sale Agreement. The relevant provisions of this Article 19 shall apply to the Drag Shareholder Instruments and references to the "Drag Shares" shall be construed accordingly.
- 19.14 Any transfer of Ordinary Shares made pursuant to, and in accordance with, this Article 19 (including the transfer of Class C Ordinary Shares pursuant to the proposed Drag Sale under Article 19.1) shall not be subject to any other restrictions on Disposal (other than those contained in Article 18 and any relating to the Lock-Up Period).
- 19.15 The Management Trustee will execute and deliver such documents within such time frame as the Majority Investors reasonably require in order to transfer the legal title to any Drag Shares held by it as nominee for a Wider Manager and which are being sold pursuant to this Article 19 including, if a share certificate has been issued in respect of any Drag Shares registered in its name, the delivery of the relevant share certificate(s) (or an indemnity in respect thereof in a form satisfactory to the Board of Managers).
- 19.16 Notwithstanding any provision to the contrary in this Article 19, no Dragged Seller who is a Manager Party shall be required to accept any shares or other securities in a company incorporated in a jurisdiction different to the jurisdiction of incorporation of the Company as consideration for any of his Drag Shares, other than where the consideration is in the form set out in Article 19.3.1 (in which case the Proposed Buyer, the Proposed Seller and the Dragged Seller shall use reasonable endeavours to implement a sale structure which is tax efficient for all parties).

20. Tag along rights.

- 20.1 If, following the end of the Lock-Up Period, a Tag Sale is proposed, no transfer of shares pursuant to the Tag Sale may be made unless:
 - 20.1.1 a Drag Notice is served in accordance with Article 19; or
- 20.1.2 the Proposed Tag Buyer makes an offer (the "Tag Offer") in writing to the Company as agent for and on behalf of:
- (a) in the case of any Tag Sale (pursuant to which less than 100% of the Class C Ordinary Shares would be transferred), the holders of Class B Ordinary Shares; or
- (b) in the case of any other Tag Sale (pursuant to which 100% of the Class C Ordinary Shares would be Disposed of) only, the holders of Class B Ordinary Shares and Class A Ordinary Shares,
 - other than the Proposed Seller(s) (the "Tag Beneficiaries") to buy:
- (i) in the case of Article 20.1.2 (a), the same proportion of the Class B Ordinary Shares held by each of the Tag Beneficiaries (together with any Class B Ordinary Shares which may be allotted in the period during which the Tag Offer is open for acceptance (the "Tag Offer Period") or, upon the Tag Offer becoming unconditional, pursuant to the exercise or conversion of options over, or rights to subscribe for securities convertible into, Class B Ordinary Shares which, in each case, were in existence at the date of the Tag Offer (together the "Tag Securities")) as the proportion of the Class C Ordinary Shares to be transferred by the Proposed Tag Seller(s) bears to the total number of Class C Ordinary Shares held by the Proposed Tag Seller(s) prior to the transfer (the "Relevant Proportion"); or
- (ii) in the case of Article 20.1.2 (b), all of the Class A Ordinary Shares and Class B Ordinary Shares held by each of the Tag Beneficiaries together with any Tag Securities (and, in the case of Article 20.1.2 (b), references to "Class B Ordinary Shares" in the definition of Tag Securities shall be construed as references to "Class A Ordinary Shares and Class B Ordinary Shares"),
 - on the terms set out in this Article 20 and the Tag Offer is or has become wholly unconditional.
 - 20.2 The terms of the Tag Offer shall be that:
- 20.2.1 it shall be open for acceptance for not less than 10 Business Days (or such lesser number of days as is agreed in writing by the Majority Investors and the Managers' Representative), and shall be deemed to have been rejected if not accepted in accordance with the terms of the Tag Offer within the Tag Offer Period;
 - 20.2.2 any acceptance of the Tag Offer shall be irrevocable;
 - 20.2.3 subject to Article 20.2.4, the consideration for each Tag Security shall be equal to the highest of:
 - (a) the consideration offered for each Class C Ordinary Share pursuant to the Tag Sale; or
- (b) the highest price per Tag Security paid by the Proposed Tag Buyer (or any party with whom the Proposed Tag Buyer is acting in concert) in the 6 months prior to the date of the Tag Offer;
- 20.2.4 notwithstanding Article 20.2.3, if the exercise of rights under Article 20.1 takes place after the process set out in Article 18 has been followed then the consideration for each Tag Security shall be equal to the average price per share for the Class C Ordinary Shares to be sold under the Tag Sale;



- 20.2.5 subject to Article 20.3, the consideration offered in respect of the Tag Securities shall be in the same form as that offered for the Class C Ordinary Shares pursuant to the Tag Sale and shall be paid at the same time and shall be subject to the same payment terms; and
 - 20.2.6 each Tagging Shareholder (as defined in Article 20.5):
- (a) shall pay its pro rata share (as a deduction from the gross pretax proceeds to be received, without prejudice to any other deductions lawfully required to be made) of the third party costs incurred by the Proposed Tag Seller(s) in connection with the Tag Sale and the transfer of the Tag Securities to the extent that such costs have been incurred on behalf of the Proposed Tag Seller(s) and all of the Tagging Shareholders; and
- (b) agrees that, in order to accept the Tag Offer, it will be required (pursuant to Article 20.8) to transfer the legal and beneficial title to its Tag Transfer Shares (as defined in Article 20.5) together with all rights attaching to them, free from all Encumbrances and with full title guarantee, and that it may also be required to give such other warranties, indemnities, covenants and undertakings as are agreed to by the Proposed Seller(s) and which the Proposed Tag Seller(s) itself or themselves shall give pursuant to the Tag Sale provided that any potential liability thereunder shall be several.
- 20.3 For the purposes of Articles 20.2, 20.8 and 20.9, "consideration" shall (unless the Majority Investors and the Managers' Representative agree otherwise):
- 20.3.1 exclude any offer to subscribe for or acquire any share, debt instrument or other security in the capital of any member of the Proposed Tag Buyer Group made to an Ordinary Shareholder provided that such offer is an alternative (whether in whole or in part) to the consideration offered for each Class C Ordinary Share under the terms of the Tag Sale: and
- 20.3.2 for the avoidance of doubt, exclude any right offered to an Ordinary Shareholder to subscribe for or acquire any share, debt instrument or other security in the capital of any member of the Proposed Tag Buyer Group in addition to the consideration offered for each Class C Ordinary Share pursuant to the Tag Sale.
- 20.4 The Tag Offer may be conditional on acceptances which would, if the relevant transfers were registered, result in the Proposed Tag Buyer Group holding or increasing its aggregate shareholding in the Company to a specified number or proportion of the Class C Ordinary Shares and/or Class A Ordinary Shares and Class B Ordinary Shares in issue. If the relevant condition is not satisfied (taking into account any Ordinary Shares transferred or proposed to be transferred pursuant to Article 20.6) or waived by the Proposed Tag Buyer, no Shares may be transferred pursuant to this Article 20 (including the Class C Ordinary Shares whose proposed transfer led to the Tag Offer).
- 20.5 The Company shall notify the holders of Tag Securities of the terms of the Tag Offer promptly upon receiving notice of the same from the Proposed Tag Buyer Group, following which any such holder who wishes to transfer Tag Securities (which, in the case of Article 20.1.2 (b), shall be all of their Tag Securities) to the Proposed Tag Buyer Group pursuant to the Tag Offer (a "Tagging Shareholder") shall serve notice on the Company to that effect (the "Tag Notice") at any time before the Tag Offer Period closes (the "Tag Closing Date") (in the case of Article 20.1.2 (a) only, stating the number of Tag Securities it wishes to transfer, which may be some or all of (but which may not exceed) the Relevant Proportion of its Tag Securities) (the "Tag Transfer Shares").
- 20.6 In the case of Article 20.1.2 (a) only, if the total number of Tag Transfer Shares set out in all Tag Notices is less than the total number of Tag Securities subject to the Tag Offer (the difference between the total number of Tag Transfer Shares set out in all Tag Notices and the total number of Tag Securities subject to the Tag Offer being the "Tag Deficit"), the Company shall notify the Proposed Tag Seller(s) who shall be entitled (but not obliged) to transfer up to such number of Class C Ordinary Shares as equals the Tag Deficit in addition to the Class C Ordinary Shares proposed to be sold by the Proposed Tag Seller(s) pursuant to the Tag Sale.
 - 20.7 On or before the date which is three days after the Tag Closing Date:
- 20.7.1 the Company shall notify the Proposed Tag Buyer Group in writing of the names and addresses of the Tagging Shareholders who have accepted the Tag Offer and the number and class of Ordinary Shares which each Tagging Shareholder wishes to transfer pursuant to the Tag Offer;
- 20.7.2 the Company shall notify each Tagging Shareholder in writing of the identity of the relevant member(s) of the Proposed Tag Buyer Group to which it shall transfer its Tag Transfer Shares; and
- 20.7.3 each of the Company's notifications above shall indicate the date, time and place on which the sale and purchase of the Tag Transfer Shares is to be completed being a date notified by the Proposed Tag Buyer Group which is not less than 7 days and not more than 14 days after the Tag Closing Date or such other date as the Majority Investors and the Proposed Tag Buyer Group may agree (the "Tag Completion Date").
- 20.8 Each Tagging Shareholder shall transfer the legal and beneficial title to its Tag Transfer Shares to the relevant member of the Proposed Tag Buyer Group on the terms set out in this Article 20, by delivering to the Company on or before the Tag Completion Date:
- 20.8.1 if a certificate has been issued in respect of any of its Tag Transfer Shares, the relevant share certificate(s) (or an indemnity in respect thereof in a form satisfactory to the Board of Managers); and
- 20.8.2 a duly executed sale agreement or form of acceptance in a form agreed by the Proposed Tag Seller(s) provided that the terms and conditions that apply to the Proposed Tag Seller(s) are not better than those applying to each Tagging Shareholder, in accordance with Article 20.2.6 (b),



and, to the extent required by the Proposed Tag Seller(s), shall sign such other documents as are signed by the Proposed Tag Seller(s) pursuant to the Tag Sale provided that the terms and conditions that apply to the Proposed Tag Seller(s) are not better than those applying to each Tagging Shareholder in such other documents, all against payment on the Tag Completion Date of the aggregate consideration due to it under the Tag Offer.

20.9 If the Proposed Tag Buyer Group has also agreed to purchase any Shareholder Instruments from the Proposed Tag Seller(s) pursuant to a Tag Sale, to the extent that some or all of the Tag Beneficiaries hold those Shareholder Instruments, the Tag Offer must also include an offer to acquire (at such consideration per relevant Shareholder Instrument as is equal to the highest consideration per relevant Shareholder Instrument offered to the Proposed Tag Seller(s) pursuant to the Tag Sale) such proportion of the relevant Shareholder Instruments held by the relevant Tag Beneficiaries as the proportion of relevant Shareholder Instruments to be transferred by the Proposed Tag Seller(s) bears to the total number of relevant Shareholder Instruments held by the Proposed Tag Seller(s) prior to the transfer. The relevant provisions of this Article 20 shall apply to the Shareholder Instruments held by the Tagging Shareholders and references to the "Tag Securities" and the "Tag Transfer Shares" shall be construed accordingly.

20.10 Any transfer of Ordinary Shares made pursuant to, and in accordance with, this Article 20 (including the transfer of Class C Ordinary Shares pursuant to a Tag Sale) shall not be subject to any other restrictions on Disposal (other than those contained in Article 18 and any relating to the Lock-Up Period).

20.11 The Management Trustee will execute and deliver such documents within such time frame as the Majority Investors reasonably require in order to transfer the legal title to any Tag Shares held by it as nominee for a Wider Manager and which are being sold pursuant to this Article 20 including, if a share certificate has been issued in respect of any Tag Securities registered in its name, the delivery of the relevant share certificate(s) (or an indemnity in respect thereof in a form satisfactory to the Board of Managers).

21. Interpretation.

- 21.1 For the purposes of Article 20 the following terms have the meanings set out below:
- 21.2 "Tag Sale" means the transfer (whether through a single transaction or a series of related transactions) of Class C Ordinary Shares by a person or persons (the "Proposed Tag Seller(s)") to a person other than to a person (the "Proposed Tag Buyer"):
 - 21.2.1 who is a connected person or an Affiliate of the Proposed Tag Seller; or
 - 21.2.2 pursuant to the Syndication Transfer Process.

22. Compulsory transfers.

- 22.1 This Article 22 applies when an employee or director of, or consultant to, any Group Company who:
- 22.1.1 is a Class A Ordinary Shareholder; or
- 22.1.2 has Disposed of Class A Ordinary Shares in accordance with Article 17.4;

ceases for any reason to be an employee or director of, or consultant to, any Group Company, such person being a "Leaver".

- 22.2 In the 12 months immediately following the relevant Cessation Date for that Leaver, the Remuneration Committee may (with Investor Consent) serve notice in writing (a "Compulsory Transfer Notice") on each or any of:
 - 22.2.1 the Class A Ordinary Shareholder who is a Leaver;
- 22.2.2 any Class A Ordinary Shareholder to whom Class A Ordinary Shares relating to that Leaver have been Disposed of under Article 17.4;
- 22.2.3 if the Leaver has died, his personal representatives and/or any other person who becomes beneficially entitled to Class A Ordinary Shares on the death of that Leaver;
- 22.2.4 if the Leaver has become bankrupt, any person who becomes entitled to Class A Ordinary Shares on his bankruptcy; and
- 22.2.5 any Class A Ordinary Shareholder who is a nominee of, or who otherwise holds Class A Ordinary Shares on behalf of, any person referred to in Articles 22.2.1 to 22.2.4 (inclusive),
 - (each a "Compulsory Transferor" and one or more of them, the "Compulsory Transferor(s)").
- 22.3 A Compulsory Transfer Notice may require the Compulsory Transferor(s) to transfer some or all of the relevant Compulsory Transfer Shares on the terms set out in this Article 22 to such person(s) as are nominated by the Remuneration Committee (with Investor Consent) from the following list:
 - 22.3.1 a replacement for any Leaver (whether internal promotion or lateral hire);
 - 22.3.2 any Manager who is a holder of Class B Ordinary Shares and is not a Leaver at that time;
- 22.3.3 another director, officer or employee of, or consultant to, a Group Company (other than to an Investor Director or a non executive director);
 - 22.3.4 to the Company in accordance with the 1915 Law;
 - 22.3.5 to the Management Trustee; or
- 22.3.6 to any warehouse, nominee, trustee or custodian (pending nomination of any of the persons referred to in Articles 22.3.1 to 22.3.4 (inclusive)), and, if no such person is nominated before an Exit occurs, to any holders of Class



B Ordinary Shares and/or Class C Ordinary Shares, pro rata to their holding of Class B Ordinary Shares and Class C Ordinary Shares, by as tax efficient a mechanism as can reasonably and cost effectively be established by the parties,

(each a "Compulsory Transferee" and one or more of them, the "Compulsory Transferee(s)") and in the case of more than one Compulsory Transferee, in the proportions indicated in the Compulsory Transfer Notice. The Compulsory Transfer Notice may reserve the right to finalise the identity of the Compulsory Transferee(s) once the price for the Compulsory Transfer Shares has been agreed or certified in accordance with Article 22.5.

- 22.4 The price for each Compulsory Transfer Share (the "Compulsory Transfer Price") shall be the price agreed in writing between the Compulsory Transferor(s) and the Remuneration Committee (with Investor Consent) or, if no such agreement is reached on or before the date which is 7 days after the date of the Compulsory Transfer Notice:
 - 22.4.1 if the Leaver is a Good Leaver, the Market Value as at the Cessation Date;
 - 22.4.2 if the Leaver is a Bad Leaver, the lower of:
- (a) the Issue Price (or where the Compulsory Transfer Shares were originally acquired by the Leaver by way of an arm's length transfer rather than allotment, the amount paid by such Leaver); and
 - (b) the Market Value as at the Cessation Date; or
 - 22.4.3 if the Leaver is an Intermediate Leaver:
 - (a) in respect of Vested Compulsory Transfer Shares, the Market Value as at the Cessation Date; or
 - (b) in respect of Unvested Compulsory Transfer Shares, the lower of:
- (i) the Issue Price (or where the Compulsory Transfer Shares were originally acquired by the Leaver by way of an arm's length transfer rather than allotment, the amount paid by such Leaver); and
 - (ii) the Market Value as at the Cessation Date.

For the purposes of this Article 22:

"C" means the Compulsory Transfer Shares;

"P" shall be calculated as follows:

= (total number of days from (and excluding) the Completion Date (or, if later, the date of issue or transfer of the Compulsory Transfer Shares to that Leaver) to (and including) the Cessation Date)/(4 x 365);

"Unvested Compulsory Transfer Shares" shall be calculated as follows:

C x X:

"Vested Compulsory Transfer Shares" shall be calculated as follows:

 $C \times P$

provided that the number of Vested Compulsory Transfer Shares shall never be less than zero; and

"X" shall be calculated as follows:

- = 1 ((total number of days from (and excluding) the Completion Date (or, if later, the date of issue or transfer of the Compulsory Transfer Shares to that Leaver) to (and including) the Cessation Date)/ (4×365)).
- 22.5 Promptly following the elapsing of the 7 days period referred to in Article 22.4, any member of the Topco Board shall instruct the Auditors (or, if the Auditors are unable or unwilling to act for any reason or the Topco Board determines that it is inappropriate for them to act, the Independent Expert) to certify the Compulsory Transfer Price as soon as possible and their decision shall (in the absence of fraud or manifest error) be final and binding on the parties. The members of the Topco Board (with an Investor Consent) are authorised to engage the Auditors/the Independent Expert jointly on behalf of themselves and the relevant Compulsory Transferor(s) and to agree the Auditors'/Independent Expert's engagement letter (on customary terms for those purposes) on behalf of themselves and the relevant Compulsory Transferor (s), and thereafter any member of the Topco Board is authorised to execute and deliver the Auditors'/Independent Expert's engagement letter for and on behalf of the Board of Managers and the relevant Compulsory Transferor(s). The costs of the Auditors (or the Independent Expert, as relevant) shall be paid by the Company unless such arrangement would not be permitted by the 1915 Law.
- 22.6 On or before the date which is 7 days after the Compulsory Transfer Price being certified under Article 22.4, the Remuneration Committee shall notify:
- 22.6.1 each Compulsory Transferor of the name(s) and address(es) of the Compulsory Transferee(s) and the number of Compulsory Transfer Shares to be transferred to each such Compulsory Transferee and the date on which the sale and purchase of the Compulsory Transfer Shares is to be completed (the "Compulsory Transfer Completion Date"), such date to be no earlier than 10 Business Days after the notification; and
 - 22.6.2 each Compulsory Transferee, indicating:
 - (a) the number of Compulsory Transfer Shares to be transferred;
 - (b) the Compulsory Transfer Price; and
 - (c) the Compulsory Transfer Completion Date.
- 22.7 The Compulsory Transferor(s) shall transfer the legal and beneficial title to the relevant Compulsory Transfer Shares to the relevant Compulsory Transferee(s) on the terms set out in this Article 22, by delivering to the Company on or before the Compulsory Transfer Completion Date:



- 22.7.1 if a certificate has been issued, the relevant share certificate(s) (or an indemnity in respect thereof in a form satisfactory to the Topco Board); and
- 22.7.2 a duly executed short form sale and purchase agreement in a form agreed with the Remuneration Committee (acting reasonably) under which the Compulsory Transferor(s) will transfer the legal and beneficial title to the relevant Compulsory Transfer Shares to the relevant Compulsory Transferee(s) free from all Encumbrances and with full title guarantee only,

against payment of the aggregate Compulsory Transfer Price due to it in cash on the Compulsory Transfer Completion Date.

- 22.8 If a Compulsory Transferor fails to comply with its obligations under Article 22.7 (a "Defaulting Compulsory Transferor"), any member of the Topco Board is authorised to execute, complete and deliver as agent for and on behalf of that Compulsory Transferor each of the documents referred to in Articles 22.7.1 and 22.7.2 (inclusive). The Board of Managers shall register the transfer(s), after which the validity of such transfer(s) shall not be questioned by any person.
- 22.9 If a certificate has been issued in respect of Compulsory Transfer Shares (or any of them), each Defaulting Compulsory Transferor shall surrender its share certificate(s) relating to the relevant Compulsory Transfer Shares (or provide an indemnity in respect thereof in a form satisfactory to the Topco Board) to the Company. On, but not before, such surrender or provision, the Defaulting Compulsory Transferor(s) shall be entitled to the aggregate Compulsory Transfer Price for the relevant Compulsory Transfer Shares transferred on its/their behalf, without interest. If a certificate has not been issued in respect of its Compulsory Transfer Shares (or any of them) the Defaulting Compulsory Transferor will be entitled, upon registration of the transfer of the Compulsory Transfer Shares pursuant to Article 22.7 (but not before) to the aggregate Compulsory Transfer Price for the relevant Compulsory Transfer Shares transferred on its/their behalf, without interest. Payment to the Compulsory Transferor(s) shall be made in such manner as is agreed between the Topco Board and the Compulsory Transferor(s) and in the absence of such agreement, by cheque to the relevant Compulsory Transferor's last known address. Receipt of the aggregate Compulsory Transfer Price for the Compulsory Transferor(s) in favour of the Compulsory Transferee(s) that the legal and beneficial title to the relevant Compulsory Transfer Shares was transferred to the Compulsory Transferee(s) free from all Encumbrances and with full title guarantee.
- 22.10 The Class A Ordinary Shareholders acknowledge and agree that the authority conferred under Article 22.8 is necessary as security for the performance by the Compulsory Transferor(s) of their obligations under this Article 22.
- 22.11 Subject to Article 22.12, a Compulsory Transferor undertakes not to attend or vote at any general meeting of the Company or (subject to the 1915 Law) at any meeting of the holders of any class of shares in the capital of the Company or for the purposes of a written resolution of the Company with automatic effect, irrespective of whether a Compulsory Transfer Notice has been served on that Compulsory Transferor pursuant to Article 22.1, and consent to not receiving notice of any such meeting and to its Compulsory Transfer Shares not being counted in determining the total number of votes which may be cast at any such meeting or required for the purposes of a written resolution or for the purposes of any other consent required under these Articles. Such Compulsory Transferors will also cease to hold any rights to receive information from the Company which they might otherwise hold pursuant to any shareholders' agreement in relation to the Company entered into from time to time.
- 22.12 The rights referred to in Article 22.11 shall be restored immediately upon the Company registering a transfer of the Compulsory Transfer Shares in accordance with this Article 22.
- 22.13 Notwithstanding Articles 22.11 and 22.12, each Compulsory Transfer Share shall at all times be subject to the provisions of Article 19.
- 22.14 No Class A Ordinary Shares or Shareholder Instruments held by a Compulsory Transferor (for the avoidance of doubt, whether or not such person has been served with a Compulsory Transfer Notice) shall be Disposed of (other than pursuant to Article 22):
 - 22.14.1 until the relevant Compulsory Transferor(s) can no longer be bound to transfer them under Article 22.3; or
- 22.14.2 without consent of the Remuneration Committee (which consent may be granted unconditionally or subject to terms or conditions, including that the Disposal be to existing Class A Ordinary Shareholders or Shareholder Instrument holders, as appropriate, on pre-emptive terms).
- 22.15 If a Tax Liability would arise on the sale of Class A Ordinary Shares by a Manager pursuant to this Article 22, the proposed buyer shall withhold such amount from the sale proceeds as will ensure that the relevant Group Company receives the amount required to discharge the Tax Liability which arises on the sale (unless the Manager agrees to fund the Tax Liability in a different way which is acceptable to the relevant Group Company).

23. Option to transfer shares.

23.1 Subject to Article 23.2 below, the Majority Investors and the Majority Managers (acting together) shall have the right (the "Option") from time to time to require all Shareholders (including the Majority Investors and the Majority Managers) (the "Option Sellers") to sell (free from all Encumbrances and any rights exercisable by third parties, and with full title guarantee) and the Transferee(s) to purchase, such number of the Ordinary Shares ("Ordinary Option Shares") and Preference Shares ("Preference Option Shares") (together the "Option Shares") as they may notify to the Option



Sellers and the Transferee(s) at a price determined pursuant to Article 23.3 below (such sale and purchase being an "Option Sale").

- 23.2 The Option may be exercisable at any time by delivery of the Option Notice, but shall be conditional upon:
- 23.2.1 the Transferee(s) having executed and delivered, prior to exercise of the Option, a deed of adherence in the form set out in any shareholders' agreement in relation to the Company entered into from time to time in which he is treated as a Transferee and having thereby become a party to that agreement;
 - 23.2.2 Majority Shareholder Consent to its exercise having been received;
- 23.2.3 the number of Option Shares to be purchased from each of the Option Sellers being pro rata to their holding of the Ordinary Shares or Preference Shares (as applicable) in issue immediately prior to the exercise of the Option; and
- 23.2.4 the Option Shares to be purchased from each of the Option Sellers being appointed between Ordinary Shares and Preference Shares pro rata with holdings of Ordinary Shares and Preference Shares of the relevant Option Seller immediately prior to the exercise of the Option.
- 23.3 The total price to be paid by the Transferee(s) for the Option Shares (the "Option Price") shall be determined by the Board of Managers, subject to Majority Shareholder Consent, it being the intention that the price payable for each of the Option Shares should be its fair market value.
 - 23.4 The Option Price shall be apportioned among the Option Shares on the same basis as is set out in Article 6.
- 23.5 Payment of the Option Price will be deferred following the transfer of the Option Share to the Transferee pursuant to the Option Sale and will be made promptly following the redemption of the Option Shares in accordance with the Articles. The Transferee is authorised to deduct and retain the Commission from the Option Price on the basis that the Commission shall be apportioned between the Option Sellers pro rata the Option Price payable to them pursuant to the Option Sale.
- 23.6 Promptly following an Option Sale, the Option Shares will be redeemed at the Option Price in accordance with the Articles.
- 23.7 Each of the Option Sellers undertakes to execute and deliver such documents, within such time frame as an Investor Director and Manager Director (acting together) may reasonably require in order to transfer legal and beneficial title to any Option Shares being sold by it to the Transferee(s) pursuant to the exercise of the Option in accordance with this Article 23, including the delivery of any share certificates (where issued) and to carry out all actions necessary or in the opinion of an Investor Director and Manager Director (acting together) desirable to effect completion of the transfer. If any Option Seller fails to comply with this Article 23.7, an Investor Director and Manager Director (acting together) may authorise any member of the Board of Managers to execute, complete and deliver as agent for and on behalf of that Option Seller any such documents. Each party will promptly take such steps and execute such documentation as may be required by an Investor Director and Manager Director (acting together) for the purposes of the redemption referred to in Article 23.6.
- 23.8 It is the intention of the Majority Investors and Majority Managers only to exercise the Option and deliver the Option Notice if they reasonably believe that the redemption of the Option Shares under Article 23.5 will take place shortly thereafter. The Shareholders shall procure that such redemption takes place as soon as reasonably practicable after the delivery of an Option Notice.
- 23.9 For the purposes of this Article 23, each Wider Manager will be deemed to be the registered holder of those shares held by the Management Trustee as nominee for that Wider Manager under the terms of the Management Trust Deed.
 - 23.10 The Management Trustee will be bound by Article 23.6 notwithstanding the provision of Article 23.7.
- 23.11 Notwithstanding any other provision of these Articles to the contrary, the Investor and the Company hereby confirm and agree that no Manager Party shall be required to accept any consideration in the form of shares or other securities in a company incorporated in a jurisdiction different to the jurisdiction of incorporation of the Company on an Option Sale, other than where the consideration is in the form set out in Article 19.3.1 (in which case the parties shall use reasonable endeavours to implement a sale structure which is tax efficient for all parties).

24. Management.

- 24.1 The Company will be managed by one or more managers who shall be appointed by a Shareholders' Resolution passed in accordance with Luxembourg Law and these Articles.
 - 24.2 If the Company has at the relevant time only one manager, he is referred to in these Articles as a "Sole Manager".
- 24.3 In case of plurality of managers (the "Board Members"), they will constitute a board of managers or conseil de gérance (the "Board of Managers") and will be divided into 2 categories, respectively denominated "Category A Managers", being the Luxembourg based managers, and "Category B Managers", being the non-Luxembourg based managers (without prejudice to other designations of the managers used in the Articles).
- 24.4 The Majority Investors are entitled from time to time to nominate for appointment to and removal from the Board of Managers of the Company up to 4 Board Members (each to be designated as a "Luxembourg Investor Director"), whether executive or non-executive, including the chairman appointed pursuant to Article 24.8.



- 24.5 The Majority Managers are entitled from time to time to nominate for appointment to and removal from the Board of Managers of the Company up to 2 Board Members (each to be designated as a "Luxembourg Manager Director"), whether executive or non-executive.
- 24.6 The Majority Investors are entitled from time to time to nominate for appointment to and removal from the Board of Managers of the Company such number of managers being ordinarily resident in Luxembourg as would constitute a majority of the members of the Board of Managers.
- 24.7 A Board Member may be removed at any time for any reason by a Shareholders' Resolution passed in accordance with Luxembourg Law and these Articles.
- 24.8 Without prejudice to any other rights that the Investors may have, the Majority Investors are entitled from time to time to nominate for appointment any Investor Director as the chairman of the Board of Managers. Appointments, removals and nominations of the chairman shall be made by notice in writing to Topco which shall, to the extent permitted by applicable law, take effect immediately upon receipt of the notice by Topco.
- **25. Powers of the managers.** The Sole Manager, when the Company has only one manager, and at all other times the Board of Managers, may take all or any action which is necessary or useful to realise any of the objects of the Company, with the exception of those reserved by Luxembourg Law or these Articles to be decided upon by the Shareholders.
- **26. Representation.** Subject as provided by Luxembourg law, these Articles and any shareholders' agreement in relation to the Company entered into from time to time, the Company is validly bound or represented towards third parties by:
 - 26.1 if the Company has a Sole Manager, the sole signature of the Sole Manager;
- 26.2 if the Company has more than one manager, the joint signature of any 2 managers, whereof 1 is a Category A Manager and 1 is a Category B Manager, at least 1 of them being a Luxembourg Investor Director; or
 - 26.3 the sole signature of any person to whom such power has been delegated in accordance with Article 27.
- 27. Agent of the managers. Subject to any shareholders' agreement in relation to the Company entered into from time to time, the Sole Manager or, if the Company has more than one manager, any 2 managers, whereof 1 is a Category A Manager and 1 is a Category B Manager, at least 1 of them being a Luxembourg Investor Director, may delegate any of their powers for specific tasks to one or more ad hoc agents and will determine any such agent's powers and responsibilities and remuneration (if any), the duration of the period of representation and any other relevant conditions of its agency.

28. Board meetings.

- 28.1 In case of plurality of managers, the meetings of the Board of Managers ("Board Meetings") are convened by (a) a majority of the Board Members of the Board of Managers, (b) a majority of those Board Members present being persons ordinarily resident in Luxembourg and (c) one Luxembourg Investor Director and, if appointed, one Luxembourg Manager Director (or one of their respective alternates), provided that, if proper notice has been given of a meeting or an adjourned meeting, a Luxembourg Manager Director need not be present. Subject to Article 24.8, the Board of Managers shall appoint the chairman.
- 28.2 A minimum of 10 Business Days' notice of meetings, accompanied by details of the venue for such meeting and an agenda of the business to be transacted and all papers to be circulated in connection with or presented to it, shall be given to all the Board Members (save in the case of an emergency, in which case such notice as is reasonably practicable in the circumstances shall be given). Unless the Luxembourg Manager Directors or the Luxembourg Investor Directors (as applicable) otherwise agree, no business shall be transacted at any such meeting except for that specified in the agenda relating to it.
- 28.3 The Board of Managers may validly debate and take decisions at a Board Meeting without complying with all or any of the convening requirements and formalities if all the Board Members have waived the relevant convening requirements and formalities either in writing or, at the relevant Board Meeting, in person or by an authorized representative.
- 28.4 A Board Member may appoint any other Board Member (but not any other person) to act as his representative at a Board Meeting to attend, deliberate, vote and perform all his functions on his behalf at that Board Meeting. A Board Member can act as representative for more than one other Board Member at a Board Meeting.
- 28.5 The Board of Managers can only validly debate and take decisions if a majority of the Board Member are present or represented, of which a majority of those Board Member present being Category A Managers and at least one Board Members being a Luxembourg Investor Director and, if appointed, one Board Members being a Luxembourg Manager Director, provided that, if proper notice has been given of a Board Meeting or an adjourned meeting, a Luxembourg Manager Director need not be present.
- 28.6 Decisions of the Board of Managers shall be adopted by a simple majority. The chairman shall be entitled to a casting vote in addition to any other vote he may have, in the case of an equality of votes on any matter before the Board of Managers.
- 28.7 A Board Members or his representative may validly participate in a Board Meeting through the medium of conference telephone, video conference or similar form of communications equipment provided that all persons participating in the meeting are able to hear and speak to each other throughout the meeting. A person participating in this way is deemed to be present in person at the meeting and shall be counted in the quorum and entitled to vote. Subject to



Luxembourg Law, all business transacted in this way by the Board Members shall, for the purposes of these Articles, be deemed to be validly and effectively transacted at a Board Meeting.

- 28.8 A resolution in writing signed by all the Board Members (or in relation to any manager, his representative) shall be as valid and effective if it had been passed at a Board Meeting duly convened and held and may consist of one or several documents in the like form each signed by or on behalf of one or more of the Board Members concerned.
- 28.9 The minutes of a meeting of the Board of Managers shall be signed by all Board Members present or represented at the meeting. Extracts shall be certified by any Board Member or by any person nominated by any Board Member or during a meeting of the Board of Managers.

29. Shareholders' resolutions.

- 29.1 Each Shareholder shall have one vote for every Share of which he is the holder.
- 29.2 Subject as provided in Articles 29.3, 29.4 and 29.5, Shareholders' Resolutions are only valid if they are passed by Shareholders holding more than half of the Shares, provided that if that figure is not reached at the first meeting or first written consultation, the Shareholders shall be convened or consulted a second time, by registered letter and the resolution may be passed by a majority of the votes cast, irrespective of the number of Shares represented.
- 29.3 Shareholders may not change the nationality of the Company or oblige any of the Shareholders to increase their participation in the Company otherwise than by unanimous vote of the Shareholders.
- 29.4 Subject as provided in Article 29.3, any resolution to change these Articles (including a change to the Registered Office), subject to any provision of the contrary, needs to be passed by a majority in number of the Shareholders representing three quarters of the Shares.
- 29.5 A resolution to dissolve the Company or to determine the method of liquidating the Company and/or to appoint the liquidators needs to be passed in accordance with Luxembourg Law.
- 29.6 A meeting of Shareholders (a "Shareholders' Meeting") may validly debate and take decisions without complying with all or any of the convening requirements and formalities if all the Shareholders have waived the relevant convening requirements and formalities either in writing or, at the relevant Shareholders' Meeting, in person or by an authorised representative.
- 29.7 A Shareholder may be represented at a Shareholders' Meeting by appointing in writing (or by fax or e-mail or any similar means) a proxy or attorney who need not be a Shareholder.
- 29.8 If at the time the Company has no more than twenty-five Shareholders, Shareholders' Resolutions may be passed by written vote of Shareholders rather than at a Shareholders' Meeting provided that each Shareholder receives the precise wording of the text of the resolutions or decisions to be adopted.
- 29.9 The majority requirement applicable to the adoption of resolutions by a Shareholders' Meeting apply mutatis mutandis to the passing of written resolutions of Shareholders. Except where required by Luxembourg Law, there shall be no quorum requirements for the passing of written resolutions of Shareholders. Written resolutions of Shareholders shall be validly passed immediately upon receipt by the Company of original copies (or copies sent by facsimile transmission or as e-mail attachments) of Shareholders' votes subject to the requirements as provided in Article 29.8 and the above provisions of Article 29.9, irrespective of whether all Shareholders have voted or not.

30. Matters requiring consent.

- 30.1 None of the Reserved Matters specified in Article 30.2 is carried out without Majority Board Consent or Majority Shareholder Consent.
 - 30.2 For the purpose of Article 30.1, the reserved matters are:
- 30.2.1 material changes to the equity or finance structure of the Group (other than share or debt issues in accordance with the equity cure and 'catch-up' rights set out in any shareholders' agreement in relation to the Company entered into from time to time, or Articles 30.2.3 and 30.2.7);
 - 30.2.2 alteration of the memorandum or articles of association of a Group Company;
- 30.2.3 dividends and other distributions or returns of capital by a Group Company that result in a Group non-share-holder debt leverage multiple beyond 4.5 times LTM EBI TDA (with EBITDA to be determined by reference to the Management Accounts at the relevant time);
 - 30.2.4 variation of share rights in respect of a Group Company;
- 30.2.5 entry into a transaction (other than the repayment of shareholder debt in accordance with its terms) by a Group Company in which any Manager Party or a director or officer (or former director or officer) of any Manager Party or a person connected with any of them would benefit;
- 30.2.6 acquisitions of or mergers with or disposals of a business or another body corporate by a Group Company, where the turnover of the target for the acquisition, merger or disposal represents more than 10 per cent. of the Group's turnover at the time of the acquisition, merger or disposal or where the equity of the Company (or any Group Company) will be issued to the vendor as part of the acquisition process;



- 30.2.7 increase in the aggregate indebtedness of the Group that result in a non-shareholder debt leverage beyond a 4.5 multiple of LTM EBI TDA (with EBI TDA determined by reference to the management accounts at the relevant time); and
- 30.2.8 any material amendment or termination of, or any agreement to materially amend or terminate, the Management Trust Deed,

together, the "Reserved Matters".

- 30.3 Subject to Article 30.7, each Manager Party undertakes for the benefit of the Investor and each of the Companies that he will procure so far as he is lawfully able that none of the acts specified in Article 30.2 is carried out without either Majority Shareholder Consent or Majority Board Consent.
- 30.4 Subject to Article 30.7, the Company undertakes for the benefit of the Investor and the Manager Parties (except to the extent that this would constitute an unlawful fetter on its statutory powers, for which purpose each of Articles 30.2.1 to 30.2.8 shall be a separate and severable undertaking by each of them), that none of the acts specified in Articles 30.2.1 to 30.2.8 will be carried out by it without either Majority Shareholder Consent or Majority Board Consent.
- 30.5 Subject to Article 30.7, the Investor undertakes for the benefit of each Manager Party and each of the Companies that it will procure so far as it is lawfully able that none of the acts specified in Articles 30.2.1 to 30.2.8 is carried out without either Majority Shareholder Consent or Majority Board Consent.
- 30.6 Subject to Article 30.7, each of the Companies severally undertakes for the benefit of the Investor and the Managers (except to the extent that this would constitute an unlawful fetter on its statutory powers, for which purpose each of Articles 30.2.1 to 30.2.8 shall be a separate and severable undertaking by it), that none of the acts specified in Articles 30.2.1 to 30.2.8 will be carried out by it and that it will procure that none of the acts specified in Articles 30.2.1 to 30.2.8 shall be carried out by any other member of the Group (except the Company) without either Majority Shareholder Consent or Majority Board Consent.
- 30.7 This Article 30 shall not apply to the extent necessary for completion of an Exit carried out in accordance with any shareholders' agreement in relation to the Company entered into from time to time, but this Article 30 does, for the avoidance of doubt, apply to a Refinancing.
- **31. Business year.** The Company's financial year starts on 1 January and ends on 31 December of each year provided that, as a transitional measure, the first financial year of the Company starts on the date of its incorporation and ends on the following 31 December (all dates inclusive).

32. Distributions on shares.

- 32.1 From the net profits of the Company determined in accordance with Luxembourg Law, five per cent (5%) shall be deducted and allocated to a legal reserve fund. That deduction will cease to be mandatory when the amount of the legal reserve fund reaches one tenth (10%) of the Company's nominal capital.
- 32.2 To the extent that funds are available at the level of the Company for distribution and to the extent permitted by law and by these Articles, the Sole Manager or in case of plurality of Board Members, the Board of Managers shall propose that cash available for remittance be distributed.
 - 32.3 Any distribution is to be made in accordance with Articles 6 to Article 13 inclusive.
- 32.4 Notwithstanding the preceding provisions, the Sole Manager or the Board of Managers as appropriate may decide to pay interim dividends to the Shareholder(s) before the end of the financial year on the basis of a statement of accounts showing that sufficient funds are available for distribution, it being understood that (i) the amount to be distributed may not exceed, where applicable, realised profits since the end of the last financial year, increased by carried forward profits and distributable reserves, but decreased by carried forward losses and sums to be allocated to a reserve to be established according to the 1915 Law or these Articles and that (ii) any such distributed sums which do not correspond to profits actually earned may be recovered from the relevant Shareholder(s).

33. Dissolution and Liquidation.

- 33.1 The Company shall not be dissolved by reason of the death, suspension of civil rights, insolvency or bankruptcy of the single Shareholder or of one of the Shareholders.
- 33.2 The liquidation of the Company shall be decided by the Shareholders' Meeting in accordance with Luxembourg Law and Article 29.
- 33.3 The liquidation will be carried out by one or several liquidators, Shareholders or not, appointed by the Shareholders who shall determine their powers and remuneration.
- 33.4 In a liquidation of the Company, the amount available for distribution shall be distributed in accordance with Articles 6 to Article 13 inclusive.

34. PECs.

- 34.1 Each party agrees that the rights of:
- 34.1.1 the Manager PEC holders to payment of any amounts in respect of par value, PIK Yield and any Unpaid PIK Yield (each as defined in the Manager PEC Instrument) on the Manager PECs; and



34.1.2 the Investor PEC holders to payment of any amount in respect of par value, Yield and any Unpaid Yield (each as defined in the Investor PEC Instrument) on the Investor PECs,

shall each be subordinated to the rights of the Manager PEC holders to receive payment of Cash Yield and Unpaid Cash Yield, in accordance with the terms of the Manager PEC Instrument and the Investor PEC Instrument.

35. Interpretation and Luxembourg law.

- 35.1 In these Articles:
- 35.1.1 a reference to:
- (a) one gender shall include each gender;
- (b) (unless the context otherwise requires) the singular shall include the plural and vice versa;
- (c) a "person" includes a reference to any individual, firm, company, corporation or other body corporate, government, state or agency of a state or any joint venture, association or partnership, works council or employee representative body (whether or not having a separate legal personality);
- (d) a statutory provision or statute includes all modifications thereto and all re-enactments (with or without modifications) thereof.
- 35.1.2 the words "include" and "including" shall be deemed to be followed by the words "without limitation" and general words shall not be given a restrictive meaning by reason of their being preceded or followed by words indicating a particular class of acts, matters or things or by examples falling within the general words;
 - 35.1.3 the headings to these Articles do not affect their interpretation or construction.
 - 35.2 In addition to these Articles, the Company is also governed by all applicable provisions of Luxembourg Law.
 - **36. Definitions.** "1915 Law" has the meaning given to it in Article 1;
 - "Acceptance Notice" has the meaning given to it in Article 18.8.2;
 - "Accepted Offeror" has the meaning given to it in Article 18.8.2;
 - "Acquisition Issue" has the meaning given to it in 5.7.4;
- "Affiliate" has the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;
 - "Articles" has the meaning given to it in Article 1;
- "Asset Sale" means a sale by the relevant member of the Group of all, or substantially all, of the Group's business, assets and undertaking, or of the share capital of a member or members of the Group carrying on and owning directly or indirectly all, or substantially all, of the Group's business, assets and undertaking;
 - "Auditors" means the auditors of the Company for the time being;
 - "Bad Leaver" means any Leaver who ceases to be employed or engaged by the Group by reason of:
- (a) his voluntary resignation, in the case of an employee or director, or by reason of his giving notice of termination of his engagement (or by his terminating his engagement without notice), in the case of a consultant; or
- (b) dismissal in circumstances justifying summary dismissal (including but not limited to fraud, cause or gross misconduct),

or any person who (even if after initially being a Good Leaver) breaches any provision of any shareholders' agreement entered into from time to time or any restrictive covenant set out in the relevant service agreement, consultancy agreement or (if applicable) any relevant compromise agreement;

"Board Meetings" has the meaning given to it in Article 28.1;

"Board Members" has the meaning given to it in Article 24.3;

"Board of Managers" has the meaning given to it in Article 24.3;

"Business Day" means a day other than a Saturday, Sunday or public holiday in London, Luxembourg, New York or lersey;

"Cash Equity Proceeds" has the meaning given to it in Article 14.1;

"Cash Yield" has the meaning given to that term in the Manager PEC Instrument;

"Cash Yield Excess" has the meaning given to it in Article 14.6.1;

"Cash Yield Shortfall" has the meaning given to it in Article 14.6.2;

"Category A Managers" has the meaning given to it in Article 24.3;

"Category B Managers" has the meaning given to it in Article 24.3;

"Cessation Date" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Class A Ordinary Shareholder" means a person entered in the register of members of the Company as the holder for the time being of a Class A Ordinary Share and "Class A Ordinary Shareholders" shall be construed accordingly;

"Class A1 Ordinary Shares" has the meaning given to it in Article 5.1.1;

"Class A2 Ordinary Shares" has the meaning given to it in Article 5.1.1;



- "Class A3 Ordinary Shares" has the meaning given to it in Article 5.1.1;
- "Class A4 Ordinary Shares" has the meaning given to it in Article 5.1.1;
- "Class A5 Ordinary Shares" has the meaning given to it in Article 5.1.1;
- "Class B Ordinary Shareholder" means a person entered in the register of members of the Company as the holder for the time being of a Class B Ordinary Share and "Class B Ordinary Shareholders" shall be construed accordingly;
 - "Class B1 Ordinary Shares" has the meaning given to it in Article 5.1.2;
 - "Class B2 Ordinary Shares" has the meaning given to it in Article 5.1.2;
 - "Class B3 Ordinary Shares" has the meaning given to it in Article 5.1.2;
 - "Class B4 Ordinary Shares" has the meaning given to it in Article 5.1.2;
 - "Class B5 Ordinary Shares" has the meaning given to it in Article 5.1.2;
- "Class C Ordinary Shareholder" means a person entered in the register of members of the Company as the holder for the time being of a Class C Ordinary Share and "Class C Ordinary Shareholders" shall be construed accordingly;
 - "Class C1 Ordinary Shares" has the meaning given to it in Article 5.1.3;
 - "Class C2 Ordinary Shares" has the meaning given to it in Article 5.1.3;
 - "Class C3 Ordinary Shares" has the meaning given to it in Article 5.1.3;
 - "Class C4 Ordinary Shares" has the meaning given to it in Article 5.1.3;
 - "Class C5 Ordinary Shares" has the meaning given to it in Article 5.1.3;
- "Commission" means an amount to be determined by the Topco Board acting reasonably as being the minimum amount practicable;
- "Companies" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;
 - "Company" has the meaning given to it in Article 1;
- "Completion" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;
 - "Completion Date" means the date of Completion;
 - "Compounded Ordinary Dividend" has the meaning given to it in Article 9.3;
 - "Compulsory Transfer Completion Date" has the meaning given to it in Article 22.6.1;
 - "Compulsory Transferee" and "Compulsory Transferee(s)" each have the meaning given to them in Article 22.3;
 - "Compulsory Transfer Notice" has the meaning given to it in Article 22.2;
 - "Compulsory Transferor" and "Compulsory Transferor(s)" each have the meaning given to them in Article 22.2;
 - "Compulsory Transfer Price" has the meaning given to it in Article 22.4;
 - "Compulsory Transfer Shares" means:
 - (a) all of the Class A Ordinary Shares held by the relevant Compulsory Transferor(s); and
- (b) any other shares in the Group held by the relevant Compulsory Transferor(s) from time to time thereafter, solely as a result of their Class A Ordinary Shareholding(s),
 - and "Compulsory Transfer Share" shall be construed accordingly;
 - "Defaulting Accepted Offeror" has the meaning given to it in Article 18.12;
 - "Defaulting Compulsory Transferor" has the meaning given to it in Article 22.8;
 - "Defaulting Dragged Seller" has the meaning given to it in Article 19.7;
- "Dispose" means, in relation to any share or any legal, beneficial or economic interest in any share or PEC, to directly or indirectly:
 - (a) sell, assign, transfer or otherwise dispose of it;
 - (b) create or permit to subsist any Encumbrance over it;
 - (c) direct (by way of renunciation or otherwise) that another person should, or assign any right to, receive it;
 - (d) enter into any agreement in respect of the votes or any other rights attached to the share; or
 - (e) agree, whether or not subject to any condition precedent or subsequent, to do any of the foregoing,
 - and "a Disposal" and "Disposed of" shall be construed accordingly;
 - "Disqualified Affiliate" has the meaning given to it in Article 17.5;
 - "Distributions" has the meaning given to it in Article 6.1.3;
 - "Dividends" has the meaning given to it in Article 6.1.3;
 - "Drag Buyer" has the meaning given to it in Article 19.1;
 - "Drag Completion Date" has the meaning given to it in Article 19.1;
 - "Dragged Seller" has the meaning given to it in Article 19.1;
 - "Drag Notice" has the meaning given to it in Article 19.1;



"Drag Price" has the meaning given to it in Article 19.1;

"Drag Shareholder Instruments" has the meaning given to it in Article 19.13;

"Drag Shares" has the meaning given to it in Article 19.1;

Eligible Ordinary Shareholder" has the meaning given to it in Article 18.2;

"Emergency Issue" has the meaning given to it in Article 5.7.4;

"Encumbrance" means a mortgage, charge, pledge, lien, option, restriction, usufruct, right of first refusal, right of preemption, third party right or interest, other encumbrance or security interest of any kind, or another type of agreement or arrangement having similar effect (and "Encumber" shall be construed accordingly);

"End Date" has the meaning given to it in Article 5.6.1;

"Equity Proceeds Apportionment" has the meaning given to it in Article 14.2;

"Excess Pre-Emption Shares" has the meaning given to it in Article 18.4.2;

"Excess Pre-Emption Shares Proportion" means, in relation to each Eligible Ordinary Shareholder who offered to acquire Excess Pre-Emption Shares, the proportion determined by dividing the number of Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable) held by that Eligible Ordinary Shareholder by the aggregate number of Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable) held by all Eligible Ordinary Shareholders who offered to buy Excess Pre-Emption Shares;

"Excess Securities" has the meaning given to it in 5.6.4;

"Exit" means:

- (a) a Sale;
- (b) a Listing; or
- (c) a Winding-Up, or
- (d) an Asset Sale; or
- a combination of some or all of them:

"Family Member" means, in relation to an A Ordinary Shareholder or a B Ordinary Shareholder, his spouse or civil partner and/or his lineal descendants by blood or adoption and/or his step-children or such other relative as agreed by the Topco Board with an Investor Consent (or, in the case of a Manager Trust, the beneficiaries of that Manager Trust);

"Family Trust" means, in relation to an A Ordinary Shareholder or a B Ordinary Shareholder, a trust (whether arising under a settlement, declaration of trust, testamentary disposition or on an intestacy) in respect of which the only beneficiaries and the only persons capable of being beneficiaries) are the A Ordinary Shareholder who is the employee or director of, or consultant to, a Group Company who established the trust and/or his Family Members, or a company wholly-owned by such a trust (or, in the case of a Manager Trust, any successor trust entity to that Manager Trust established with the same beneficiaries);

"Fifth Series Ordinary Shares" means the Class A5 Ordinary Shares, Class B5 Ordinary Shares and the Class C5 Ordinary Shares;

"First Series Ordinary Shares" means the Class A1 Ordinary Shares, Class B1 Ordinary Shares and the Class C1 Ordinary Shares;

"Fourth Series Ordinary Shares" means the Class A4 Ordinary Shares, Class B4 Ordinary Shares and the Class C4 Ordinary Shares;

"FSA" means the Financial Services Authority;

"FSMA" means the Financial Services and Markets Act 2000;

"Fund" means any unit trust, investment trust, investment company, limited partnership, general partnership, collective investment scheme, pension fund, insurance company, authorised person under FSMA or any body corporate or other entity, in each case the assets of which are managed professionally for investment purposes;

"Further Drag Completion Date" has the meaning given to it in Article 19.12;

"Further Drag Notice" has the meaning given to it in Article 19.12;

"Further Dragged Seller" has the meaning given to it in Article 19.12;

"Good Leaver" means any Leaver who ceases to be employed or engaged by the Group by reason of:

- (a) death; or
- (b) long term ill health or physical or mental disability which renders him incapable of continued employment in his current position carrying out the normal duties for that position, as certified by a general medical practitioner, or other specialist medical professional, nominated or approved by the Remuneration Committee; or
 - (c) retirement at normal retirement age in line with the Group's own retirement policy; or
 - (d) being made redundant;

"Group" (except where specifically defined otherwise) means the Companies and their subsidiary undertakings for the time being (including, for the avoidance of doubt, SBR Europe S.a r.l.) and any holding company of the Company in which



the share capital structure of the Company is replicated in all material respects, and "member of the Group" and "Group Company" shall be construed accordingly;

"Holding Entity" has the meaning given to it in Article 3.6;

"Independent Expert" means a valuation expert (acting as an expert and not as an arbitrator) nominated by the parties concerned and appointed by the Topco Board on the basis set out in Article 22.5 or, in the event of disagreement as to nomination, appointed upon application by the Topco Board by the President for the time being of the Institute of Chartered Accountants in England and Wales;

"Intermediate Leaver" means:

- (a) a Leaver who is not a Good Leaver or a Bad Leaver; or
- (b) Sundeep Arora, in the event that (i) he is a Leaver; and (ii) the Investor has changed him as the CEO pursuant to the relevant clause in the shareholders' agreement in relation to the Company entered into from time to time in circumstances where he is not a Good Leaver or a Bad Leaver;

"Investor" means any person who holds Class C Ordinary Shares;

"Investor Consent" means either an Investor Director Consent or the consent in writing of the Majority Investors;

"Investor Director" means a director appointed to the Topco Board by the Majority Investors;

"Investor Director Consent" means the consent in writing of an Investor Director;

"Investor PEC Instrument" means the instrument in the agreed form to be executed by the Company constituting the Investor PECs;

"Investor PECs" means the £333,629,716.50 12 per cent. unsecured subordinated PECs to be issued by the Company constituted by the Investor PEC Instrument and "Investor PEC" shall be construed accordingly;

"Issue Price" means the aggregate of the amount paid up (or credited as paid up) in respect of the par value, together with any amount credited to the share premium account, in respect of the relevant share in the capital of the Company;

"Leaver" has the meaning given to it in Article 22.1;

"Listing" means:

- (a) both the admission of any of the relevant Group Company's shares to the official list maintained by the FSA becoming effective (in accordance with the Listing Rules) and the admission of any of the relevant Group Company's shares to trading on the LSE's market for listed securities (in accordance with the Admission and Disclosure Standards of the LSE, for the time being in force); or
- (b) the admission to trading of any of the relevant Group Company's shares on AIM, the market operated by the LSE, becoming effective; or
- (c) the equivalent admission to trading to or permission to deal on any other Recognised Investment Exchange, or such other investment exchange as is nominated by the Majority Investors, becoming effective in relation to any of the relevant Group Company's shares;

"Listing Rules" means the rules made by the FSA pursuant to section 73A of FSMA, for the time being in force;

"Lock-Up Period" has the meaning given to it in Article 17.1;

"LSE" means the London Stock Exchange plc;

"LTM EBITDA" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Luxembourg Investor Director" has the meaning given to it in Article 24.4;

"Luxembourg Law" has the meaning given to it in Article 2.2.2;

"Luxembourg Manager Director" has the meaning given to it in Article 24.5;

"Majority Board Consent" means any decision taken at a meeting of the Topco Board by a majority of the votes of participating directors and in respect of which both an Investor Director and a Manager Director vote in favour;

"Majority Investors" means persons who hold more than 50 per cent. of the Class C Ordinary Shares for the time being in issue;

"Majority Managers" means persons who hold more than 50 per cent. of the Class B Ordinary Shares for the time being in issue;

"Majority Shareholder Consent" means:

- (a) an Investor Consent; together with
- (b) the consent in writing of the Majority Managers;

"Management Trust Deed" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Management Trustee" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Manager" means any person who holds Class A Ordinary Shares or Class B Ordinary Shares;

"Manager Director" means a director appointed to the Topco Board by the Majority Managers;



"Manager Party" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time, and "Manager Parties" shall be construed accordingly;

Manager PEC Instrument" means the instrument in the agreed form to be executed by the Company constituting the Manager PECs;

"Manager PECs" means the £222,419,810.97 12 per cent. plus 0.801 per cent. unsecured subordinated PECs to be issued by the Company constituted by the Manager PEC Instrument and "Manager PEC" shall be construed accordingly;

"Managers' Representative" means Sundeep Arora (provided he is a Class B Ordinary Shareholder) or such other person who is a Class B Ordinary Shareholder as is nominated in writing by the holders of more than 50 per cent. of the Class B Ordinary Shares for the time being in issue;

"Manager Trusts" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time, and "Manager Trust" shall be construed accordingly;

"Market Value" means, in relation to the relevant share or shares in the capital of the Company, the price which the Auditors acting as experts and not as arbitrators (or, if the Auditors are unwilling or unable to act or the Topco Board determines that it is inappropriate for them to act, the Independent Expert) state in writing to be in their opinion their market value, on the basis of a sale as between a willing seller and a willing buyer at arm's length and, in determining such market value, the Auditors/ Independent Expert shall be instructed in particular:

- (a) to have regard to the rights and restrictions attached to such shares in respect of income and capital (including the operation of the ratchet mechanism set out in Article 6) but disregard any restrictions as to transfer;
 - (b) to disregard whether such shares represent a minority or a majority interest, as appropriate;
- (c) to take full account of the fully diluted share capital of the Company and of the loan capital and debt structure of the Group; and
 - (d) to have regard to such other factors as they shall regard appropriate for such purpose;
 - "New Issue Notice" has the meaning given to it in Article 5.6.1;
 - "Nominated Account Holder" has the meaning given to it in Article 6.4.15;
 - "Nominated Bank Account" has the meaning given to it in Article 6.4.14;
 - "Non-cash Equity Proceeds" has the meaning given to it in Article 14.4;
 - "Option" has the meaning given to it in Article 23.1;

"Option Notice" means the notice provided to the Transferee(s) and the Options Sellers by the Majority Investor and Majority Mangers pursuant to Article 23.2;

"Option Price" has the meaning given to it in Article 23.3;

"Option Sale" has the meaning given to it in Article 23.1;

"Option Shares" has the meaning given to it in Article 23.1;

"Ordinary Dividend" has the meaning given to it in Article 9.2;

"Ordinary Option Shares" has the meaning given to it in Article 23.1;

"Ordinary Share Amount" means:

- (a) in respect of any First Series Ordinary Shares, its nominal value plus any accruals of Ordinary Dividend;
- (b) in respect of any Second Series Ordinary Shares, its nominal value plus any accruals of Ordinary Dividend;
- (c) in respect of any Third Series Ordinary Shares, its nominal value plus any accruals of Ordinary Dividend;
- (d) in respect of any Fourth Series Ordinary Shares, its nominal value plus any accruals of Ordinary Dividend; and
- (e) in respect of any Fifth Series Ordinary Shares, its nominal value plus any accruals of Ordinary Dividend;

"Ordinary Shareholders" means the Class A Ordinary Shareholders, the Class B Ordinary Shareholders and the Class C Ordinary Shareholders and "Ordinary Shareholder" shall be construed accordingly;

"Ordinary Shares" means the Class A Ordinary Shares, the Class B Ordinary Shares and/or the Class C Ordinary Shares (as applicable);

"Ordinary Share Series" means the First Series Ordinary Shares, Second Series Ordinary Shares, Third Series Ordinary Shares, Fourth Series Ordinary Shares and Fifth Series Ordinary Shares, as applicable;

"Other Securities" has the meaning given to it in Article 5.6.1;

"PECs" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time:

"Preference Distribution Proportions" has the meaning given to it in Article 6.4.16;

"Preference Dividend" has the meaning given to it in Article 8.2;

"Preference Option Shares" has the meaning given to it in Article 23.1;

"Preference Share Amount" has the meaning given to it in Article 6.4.18;

"Preference Shares" has the meaning given to it in Article 5.1.4;

"Pre-Emption Completion Date" has the meaning given to it in Article 18.9.1;



"Pre-Emption Notice" has the meaning given to it in Article 18.2;

"Pre-Emption Offer" has the meaning given to it in Article 18.4;

"Pre-Emption Offer Closing Date" has the meaning given to it in Article 18.4;

"Pre-Emption Offeror" has the meaning given to it in Article 18.4;

"Pre-Emption Price" has the meaning given to it in Article 18.5;

"Pre-Emption Proportion" means, in relation to each Eligible Ordinary Shareholder, the pro rata proportion determined by dividing the number of Class B Ordinary Shares and/or Class C Ordinary Shares (as applicable) held by that Eligible Ordinary Shareholder by the aggregate number of Class B Ordinary Shares and Class C Ordinary Shares held by all Class B Ordinary Shareholders and Class C Ordinary Shareholders;

"Pre-Emption Shareholder Instruments" has the meaning given to it Article 18.16;

"Pre-Emption Shares" has the meaning given to it in Article 18.2;

"Pre-Emption Transfer" has the meaning given to it in Article 18.1;

"Proposed Buyer" has the meaning given to it in the definition of "Sale";

"Proposed Buyer Group" has the meaning given to it in the definition of "Sale";

"Proposed Seller(s)" has the meaning given to it in the definition of "Sale";

"Proposed Transferor(s)" has the meaning given to it in Article 18.1;

"Ratchet Expert" has the meaning given to it in Article 6.3;

"Ratchet Relevant Date" has the meaning given to it in Article 6.4.19;

"Recognised Investment Exchange" has the meaning given to it in section 285 FSMA;

"Redemption Payments" has the meaning given to it in Article 6.1.3;

"Refinancing" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time.

"Registered Office" has the meaning given to it in Article 2.1;

"Rejection Notice" has the meaning given to it in Article 18.8.1;

"Related Holder" means a Family Member or a Family Trust;

"Relevant Proportion" has the meaning given to it in Article 19.1;

"Relevant Securities" has the meaning given to it in Article 5.6.1;

"Remuneration Committee" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Reserved Matters" has the meaning given to it in Article 30.2;

"Sale" means the transfer (whether through a single transaction or a series of related transactions) of Ordinary Shares on an arm's length basis by a person or persons (the "Proposed Seller(s)") which, if registered, would result in a person, being a bona fide unconnected offeror for the Ordinary Shares, (the "Proposed Buyer") and any other person:

- (a) who is a connected person of the Proposed Buyer; or
- (b) with whom the Proposed Buyer is acting in concert;

(together the "Proposed Buyer Group"), other than an Affiliate of the Proposed Seller(s), holding more than 50 per cent. of the Class C Ordinary Shares for the time being in issue;

"Sale Agreement" has the meaning given to it in Article 19.1;

"Sale Payments" has the meaning given to it in 6.1.1;

"Second End Date" has the meaning given to it in 5.6.4;

"Second Series Ordinary Shares" means the Class A2 Ordinary Shares, Class B2 Ordinary Shares and the Class C2 Ordinary Shares;

"Share Premium Account" has the meaning given to it in Article 5.2;

"Share Redemption" has the meaning given to it in 5.5;

"Shareholders" has the meaning given in Article 5.1 and "Shareholder" shall be construed accordingly;

"Shareholder Instruments" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time (and "Shareholder Instrument" means any one of them);

"Shareholders' Meeting" has the meaning given to it in Article 29.6;

"Shareholders' Resolution" has the meaning given to it in Article 2.2.2;

"Shares" has the meaning given in Article 5.1;

"Sole Manager" has the meaning given to it in Article 24.2;

"Subsequent Dragged Shares" has the meaning given to it in Article 19.12;

"Subsequent Redemption Period" has the meaning given to it in Article 13.3;

"Subsequent Shares" has the meaning given to it in Article 19.12;



"Syndicatee" means any person who is an investor in any Fund which is an Affiliate of the Investor from time to time;

"Syndication Transfer Process" means one or several transfers each of which is completed within 9 months of the date of Completion of an aggregate number (when added to any previous such transfer(s)) of not more than 20 per cent. of the Class C Ordinary Shares (together with such proportion of the Shareholder Instruments held by the Investor as the number of Class C Ordinary Shares so transferred bears to the total number of Class C Ordinary Shares held by the Investor);

"Tag Beneficiaries" has the meaning given to it in Article 20.1.2;

"Tag Closing Date" has the meaning given to it in Article 20.5;

"Tag Completion Date" has the meaning given to it in Article 20.7.3;

"Tag Deficit" has the meaning given to it in Article 20.6;

"Tagging Shareholder" has the meaning given to it in Article 20.5;

"Tag Notice" has the meaning given to it in Article 20.5;

"Tag Offer" has the meaning given to it in Article 20.1.2;

"Tag Offer Period" has the meaning given to it in Article 20.1.2;

"Tag Sale" has the meaning given to it in Article 21.2;

"Tag Securities" has the meaning given to it in Article 20.1.2;

"Tag Transfer Shares" has the meaning given to it in Article 20.5;

"Tax Liability" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Third Series Ordinary Shares" means the Class A3 Ordinary Shares, Class B3 Ordinary Shares and the Class C3 Ordinary Shares;

"Topco" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Topco Board" means the board of directors of Topco from time to time;

"Transferee(s)" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time;

"Wider Manager Shares" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time and "Wider Manager Share" shall be construed accordingly;

"Wider Managers" shall have the meaning given to it in any shareholders' agreement in relation to the Company entered into from time to time, and "Wider Manager" shall be construed accordingly; and

"Winding-Up" means a distribution to Shareholders of all or substantially all of the assets of the Group pursuant to a winding-up or dissolution of the Company or any holding company of the Company in which the share capital structure of the Company is replicated in all material respects.

Costs and Expenses

The expenses, costs, remunerations or charges in any form whatsoever which shall be borne by the Company as a result of the present deed are estimated at approximately two thousand eight hundred euro (EUR 2,800.-).

Nothing else being on the agenda, and nobody rising to speak, the meeting is closed.

The present deed is drawn up in Esch/Alzette, on the day named at the beginning of this document.

The undersigned notary who understands and speaks English, states herewith that at the request of the Shareholders, the present deed is written in English, followed by a French version, at the request of the same Shareholders, and in case of divergences between the English and the French texts, the English version will prevail.

The document having been read to the proxyholder of the Shareholders, known to the notary by name, first name, civil status and residence, said proxyholder signed together with the notary the present deed.

Suit la traduction française de ce qui précède:

(N.B. Pour des raisons techniques, ladite version française est publiée au Mémorial C-N° 1823 du 29 juillet 2013.)

Signé: Conde, Rouckert, Maria Santiago, Kesseler.

Enregistré à Esch/Alzette Actes Civils, le 28 mai 2013. Relation: EAC/2013/6751. Reçu soixante-quinze euros 75,00€ Le Receveur ff. (signé): M. Halsdorf.

POUR EXPEDITION CONFORME.

Référence de publication: 2013083823/1681.

(130103491) Déposé au registre de commerce et des sociétés de Luxembourg, le 26 juin 2013.

Editeur: Service Central de Législation, 43, boulevard F.-D. Roosevelt, L-2450 Luxembourg

Imprimeur: Association momentanée Imprimerie Centrale / Victor Buck